%&pdflatex 
\documentclass{amsart}
%\synctex=1
%\setlength{\paperheight}{ 220mm}
%\setlength{\paperwidth}{ 155mm}

%\usepackage[notref, notcite]{showkeys}% notref
%\usepackage[pagewise]{lineno}
%\linenumbers

%\RequirePackage{mathmacros}
%\usepackage[english,german]{babel}
%\NeedsTeXFormat{LaTeX2e}
%\ProvidesPackage{mathmacros}[2001/9/11 CG]

%\newcommand{\anl}{\htmladdnormallink}%requires hyperref
%besser: \href, requires hyperref
\RequirePackage{amsmath}
\RequirePackage{bm}
\RequirePackage{amssymb}
\RequirePackage{upref}
\RequirePackage{amsthm}
\RequirePackage{enumerate}
\RequirePackage{pb-diagram}
\RequirePackage{amsfonts}
\RequirePackage[mathscr]{eucal}
\RequirePackage{verbatim}
\RequirePackage{xr}
\RequirePackage{graphicx}
%\RequirePackage{floatflt}
\usepackage{calc}
\usepackage{xspace}
\RequirePackage{color}
\RequirePackage{ifthen}

%\def\@thm#1#2#3{%
%  \ifhmode\unskip\unskip\par\fi
%  \normalfont
%  \trivlist
%  \let\thmheadnl\relax
%  \let\thm@swap\@gobble
%  \let\thm@indent\indent % no indent
%  \thm@headfont{\scshape}% heading font bold
%  %\thm@notefont{\fontseries\mddefault\upshape}%
%  \thm@notefont{}%
%  \thm@headpunct{.}% add period after heading
%  \thm@headsep 5\p@ plus\p@ minus\p@\relax
%  \thm@preskip\topsep
%  \thm@postskip\thm@preskip
%  #1% style overrides
%  \@topsep \thm@preskip               % used by thm head
%  \@topsepadd \thm@postskip           % used by \@endparenv
%  \def\@tempa{#2}\ifx\@empty\@tempa
%    \def\@tempa{\@oparg{\@begintheorem{#3}{}}[]}%
%  \else
%    \refstepcounter{#2}%
%    \def\@tempa{\@oparg{\@begintheorem{#3}{\csname the#2\endcsname}}[]}%
%  \fi
%  \@tempa
%}

%Abbreviations

\newcommand{\cf}{cf.\@\xspace}
\newcommand{\resp}{resp.\@\xspace}
\newcommand{\ac}{a.c.\@\xspace}
\newcommand{\aev}{a.e.\@\xspace}
\newcommand{\st}{s.t.\@\xspace}
\newcommand{\wlogc}{w.l.o.g.\@\xspace}
\newcommand{\Wlogc}{W.l.o.g.\@\xspace}
\newcommand{\lsc}{l.s.c.\@\xspace}
\newcommand{\usc}{u.s.c.\@\xspace}
\newcommand{\MVT}{MVT}
\newcommand{\IVP}{IVP}
\newcommand{\iffc}{iff}
\newcommand{\Str}{Str.\@\xspace}



%Redefined commands


%Greek Letters

\newcommand{\al}{\alpha}
\newcommand{\bet}{\beta}
\newcommand{\ga}{\gamma}
\newcommand{\de}{\delta }
\newcommand{\e}{\epsilon}
\newcommand{\ve}{\varepsilon}
\newcommand{\f}{\varphi}
\newcommand{\h}{\eta}
\newcommand{\io}{\iota}
\newcommand{\tht}{\theta}
\newcommand{\ka}{\kappa}
\newcommand{\lam}{\lambda}
\newcommand{\m}{\mu}
\newcommand{\n}{\nu}
\newcommand{\om}{\omega}
\newcommand{\p}{\pi}
\newcommand{\vt}{\vartheta}
\newcommand{\vr}{\varrho}
\newcommand{\s}{\sigma}
\newcommand{\x}{\xi}
\newcommand{\z}{\zeta}

\newcommand{\C}{\varGamma}
\newcommand{\D}{\varDelta}
\newcommand{\F}{\varPhi}
\newcommand{\Lam}{\varLambda}
\newcommand{\Om}{\varOmega}
\newcommand{\vPsi}{\varPsi}
\newcommand{\Si}{\varSigma}

%New Commands

\newcommand{\di}[1]{#1\nobreakdash-\hspace{0pt}dimensional}%\di n
\newcommand{\nbdd}{\nobreakdash--}
\newcommand{\nbd}{\nobreakdash-\hspace{0pt}}
\newcommand{\ce}[1]{$C^#1$\nbd{estimate}}
\newcommand{\ces}[1]{$C^#1$\nbd{estimates}}


\newcommand{\fm}[1]{F_{|_{M_#1}}}
\newcommand{\fmo}[1]{F_{|_{#1}}}%\fmo M
\newcommand{\fu}[3]{#1\hspace{0pt}_{|_{#2_#3}}}
\newcommand{\fv}[2]{#1\hspace{0pt}_{|_{#2}}}
\newcommand{\cchi}[1]{\chi\hspace{0pt}_{_{#1}}}
\newcommand{\tchi}[1]{\tilde\chi\hspace{0pt}_{_{#1}}}
\newcommand{\so}{{\mc S_0}}
%\newcommand\sql[1][u]{\sqrt{1-|D#1|^2}}


\newcommand{\const}{\tup{const}}
\newcommand{\ndash}{\nobreakdash--}


\newcommand{\slim}[2]{\lim_{\substack{#1\ra #2\\#1\ne #2}}}


\newcommand{\pih}{\frac{\pi}{2}}


\newcommand{\msp[1]}[1]{\mspace{#1mu}}
\newcommand{\low}[1]{{\hbox{}_{#1}}}



%Special Symbols

\newcommand{\R}[1][n+1]{{\protect\mathbb R}^{#1}}
\newcommand{\Ccc}[1][n]{{\protect\mathbb C}^{#1}}
\newcommand{\Hh}[1][n+1]{{\protect\mathbb H}^{#1}}
\newcommand{\Ss}[1][n+1]{{\protect\mathbb S}^{#1}}
\newcommand{\Cc}{{\protect\mathbb C}}
\newcommand{\K}{{\protect\mathbb K}}
\newcommand{\N}{{\protect\mathbb N}}
\newcommand{\Q}{{\protect\mathbb Q}}
\newcommand{\Z}{{\protect\mathbb Z}}
\newcommand{\eR}{\stackrel{\lower1ex \hbox{\rule{6.5pt}{0.5pt}}}{\msp[3]\R[]}}
\newcommand{\eN}{\stackrel{\lower1ex \hbox{\rule{6.5pt}{0.5pt}}}{\msp[1]\N}}
\newcommand{\eO}{\stackrel{\lower1ex \hbox{\rule{6pt}{0.5pt}}}{\msc O}}
\newcommand{\mf}[1]{\mathfrak {#1}}




%Special math symbols

%\DeclareMathOperator{\arccot}{arccot}
%\DeclareMathOperator{\arccos}{arccos}
%\DeclareMathOperator{\arctan}{arctan}
\DeclareMathOperator{\diam}{diam}
\DeclareMathOperator{\Grad}{Grad}
\DeclareMathOperator*{\es}{ess\,sup}
\DeclareMathOperator{\graph}{graph}
\DeclareMathOperator{\sub}{sub}
\DeclareMathOperator{\supp}{supp}
\DeclareMathOperator{\id}{id}
\DeclareMathOperator{\lc}{lc}
\DeclareMathOperator{\osc}{osc}
\DeclareMathOperator{\pr}{pr}
\DeclareMathOperator{\rec}{Re}
\DeclareMathOperator{\imc}{Im}
\DeclareMathOperator{\sign}{sign}
\DeclareMathOperator{\proj}{proj}
\DeclareMathOperator{\grad}{grad}
\DeclareMathOperator{\Diff}{Diff}
\DeclareMathOperator{\rg}{rg}
\DeclareMathOperator{\indm}{ind}
\DeclareMathOperator{\ad}{ad}
\DeclareMathOperator{\Ad}{Ad}
\DeclareMathOperator{\Int}{Int}
\DeclareMathOperator{\Aut}{Aut}
\DeclareMathOperator{\GL}{GL}
\DeclareMathOperator{\aut}{\mf{aut}}
\DeclareMathOperator{\I}{I}
\DeclareMathOperator{\B}{B}
\DeclareMathOperator{\tr}{tr}
\DeclareMathOperator{\SO}{SO}
\DeclareMathOperator{\SU}{SU}
\DeclareMathOperator{\U}{U}
\DeclareMathOperator{\soc}{\mf{so}}
\DeclareMathOperator{\suc}{\mf{su}}
\DeclareMathOperator{\uc}{\mf{u}}
\DeclareMathOperator{\diag}{diag}
\DeclareMathOperator{\dist}{dist}
\DeclareMathOperator{\divc}{div}

\newcommand\im{\implies}
\newcommand\ra{\rightarrow}
\newcommand\xra{\xrightarrow}
\newcommand\rra{\rightrightarrows}
\newcommand\hra{\hookrightarrow}
\newcommand{\nea}{\nearrow}
\newcommand{\sea}{\searrow}
\newcommand{\ua}{\uparrow}
\newcommand{\da}{\downarrow}
\newcommand{\rha}{\rightharpoondown}
\newcommand{\wha}{\underset{w^*}\rightharpoondown}
\newcommand\whc[1]{\underset{#1}\rightharpoondown}

%PDE commands

\newcommand\pa{\partial}
\newcommand\pde[2]{\frac {\partial#1}{\partial#2}}
\newcommand\pd[3]{\frac {\partial#1}{\partial#2^#3}}   %e.g. \pd fxi
\newcommand\pdc[3]{\frac {\partial#1}{\partial#2_#3}}   %contravariant
\newcommand\pdm[4]{\frac {\partial#1}{\partial#2_#3^#4}}   %mixed
\newcommand\pdd[4]{\frac {{\partial\hskip0.15em}^2#1}{\partial {#2^
#3}\,\partial{#2^#4}}}    %e.g. \pdd fxij, Abl. zweiter Ordnung
\newcommand\pddc[4]{\frac {{\partial\hskip0.15em}^2#1}{\partial {#2_
#3}\,\partial{#2_#4}}} 
\newcommand\PD[3]{\frac {{\partial\hskip0.15em}^2#1}{\partial
#2\,\partial#3}}       %e.g \PD fxy

\newcommand\df[2]{\frac {d#1}{d#2}}
\newcommand\dde[2]{\frac {\de#1}{\de#2}}


\newcommand\sd{\vartriangle}
\newcommand\sq[1][u]{\sqrt{1+|D#1|^2}}
\newcommand\sql[1][u]{\sqrt{1-|D#1|^2}}
\newcommand{\un}{\infty}
\newcommand{\A}{\forall}
\newcommand{\E}{\exists}

%Set commands

\newcommand{\set}[2]{\{\,#1\colon #2\,\}}
\newcommand{\uu}{\cup}
\newcommand{\ii}{\cap}
\newcommand{\uuu}{\bigcup}
\newcommand{\iii}{\bigcap}
\newcommand{\uud}{ \stackrel{\lower 1ex \hbox {.}}{\uu}}
\newcommand{\uuud}[1]{ \stackrel{\lower 1ex \hbox {.}}{\uuu_{#1}}}
\newcommand\su{\subset}
\newcommand\Su{\Subset}
\newcommand\nsu{\nsubset}
\newcommand\eS{\emptyset}
\newcommand{\sminus}[1][28]{\raise 0.#1ex\hbox{$\scriptstyle\setminus$}}
\newcommand{\cpl}{\complement}

\newcommand\inn[1]{{\stackrel{\msp[9]\circ}{#1}}}



%Embellishments

\newcommand{\ol}{\overline}
\newcommand{\pri}[1]{#1^\prime}
\newcommand{\whn}[1]{\widehat{(#1_n)}}
\newcommand{\wh}{\widehat}
\newcommand{\wt}{\widetilde}


%Logical commands

\newcommand{\wed}{\wedge}
\newcommand{\eqv}{\Longleftrightarrow}
\newcommand{\lla}{\Longleftarrow}
\newcommand{\lra}{\Longrightarrow}
\newcommand{\bv}{\bigvee}
\newcommand{\bw}{\bigwedge}

\newcommand{\nim}{{\hskip2.2ex\not\hskip-1.5ex\im}}

\DeclareMathOperator*{\Au}{\A}
\DeclareMathOperator*{\Eu}{\E}

\newcommand\ti{\times }


%Norms
\newcommand{\abs}[1]{\lvert#1\rvert}
\newcommand{\absb}[1]{\Bigl|#1\Bigr|}
\newcommand{\norm}[1]{\lVert#1\rVert}
\newcommand{\normb}[1]{\Big\lVert#1\Big\rVert}
\newcommand{\nnorm}[1]{| \mspace{-2mu} |\mspace{-2mu}|#1| \mspace{-2mu}
|\mspace{-2mu}|}
\newcommand{\spd}[2]{\protect\langle #1,#2\protect\rangle}

%Geometry
\newcommand\ch[3]{\varGamma_{#1#2}^#3}
\newcommand\cha[3]{{\bar\varGamma}_{#1#2}^#3}
\newcommand\chc[3]{{\tilde\varGamma}_{#1#2}^#3}
\newcommand{\riem}[4]{R_{#1#2#3#4}}
\newcommand{\riema}[4]{{\bar R}_{#1#2#3#4}}
\newcommand{\cod}{h_{ij;k}-h_{ik;j}=\riema\al\bet\ga\de\n^\al x_i^\bet x_j^\ga x_k^\de}
\newcommand{\gau}[1][\s]{\riem ijkl=#1 \{h_{ik}h_{jl}-h_{il}h_{jk}\} + \riema
\al\bet\ga\de x_i^\al x_j^\bet x_k^\ga x_l^\de}
\newcommand{\ric}{\h_{i;jk}=\h_{i;kj}+\riem lijk\msp \h^l}
\newcommand{\rnn}{\bar R_{\al\bet}\nu^\al\nu^\bet}
\newcommand{\ein}{G_{\al\bet}\nu^\al\nu^\bet}

%Font commands

\newcommand{\tbf}{\textbf}
\newcommand{\tit}{\textit}
\newcommand{\tsl}{\textsl}

\newcommand{\tsc}{\textsc}
\newcommand{\trm}{\textrm}
\newcommand{\tup}{\textup}% text upright

\newcommand{\mbf}{\protect\mathbf}
\newcommand{\mitc}{\protect\mathit}
\newcommand{\mrm}{\protect\mathrm}


\newcommand{\bs}{\protect\boldsymbol}
\newcommand{\mc}{\protect\mathcal}
\newcommand{\msc}{\protect\mathscr}



%Miscellaneous


\newcommand\mbb[1]{{{\hfill\bf\boxed{#1}}}}
\providecommand{\bysame}{\makebox[3em]{\hrulefill}\thinspace}
\newcommand{\la}{\label}
\newcommand{\ci}{\cite}
\newcommand{\bib}{\bibitem}

\newcommand{\cq}[1]{\glqq{#1}\grqq\,}
\newcommand{\cqs}[1]{\glqq{\,#1\,}\grqq\,}
\newcommand{\cqp}[1]{\glqq{\ignorespaces #1\ignorespaces}\grqq}
\newcommand{\cqr}{\glqq{$\lra$}\grqq\,}
\newcommand{\cql}{\glqq{$\lla$}\grqq\,}

\newcommand\fr[2]{\frac{#1}{1-\frac14{#2}^2}}
\newcommand\ffr[2]{\frac{#1}{(1-\frac14{#2}^2)^2}}

\newcommand{\bt}{\begin{thm}}
\newcommand{\bl}{\begin{lem}}
\newcommand{\bc}{\begin{cor}}
\newcommand{\bd}{\begin{definition}}
\newcommand{\bpp}{\begin{prop}}
\newcommand{\br}{\begin{rem}}
\newcommand{\bn}{\begin{note}}
\newcommand{\be}{\begin{ex}}
\newcommand{\bes}{\begin{exs}}
\newcommand{\bb}{\begin{example}}
\newcommand{\bbs}{\begin{examples}}
\newcommand{\ba}{\begin{axiom}}
\newcommand{\bas}{\begin{assumption}}



\newcommand{\et}{\end{thm}}
\newcommand{\el}{\end{lem}}
\newcommand{\ec}{\end{cor}}
\newcommand{\ed}{\end{definition}}
\newcommand{\epp}{\end{prop}}
\newcommand{\er}{\end{rem}}
\newcommand{\en}{\end{note}}
\newcommand{\ee}{\end{ex}}
\newcommand{\ees}{\end{exs}}
\newcommand{\eb}{\end{example}}
\newcommand{\ebs}{\end{examples}}
\newcommand{\ea}{\end{axiom}}
\newcommand{\eas}{\end{assumption}}


\newcommand{\bp}{\begin{proof}}
\newcommand{\ep}{\end{proof}}
\newcommand{\eps}{\renewcommand{\qed}{}\end{proof}}

\newcommand{\bal}{\begin{align}}
%\newcommand{\eal}{\end{align}}


\newcommand{\bi}[1][1.]{\begin{enumerate}[\upshape #1]}
\newcommand{\bia}[1][(1)]{\begin{enumerate}[\upshape #1]}
\newcommand{\bin}[1][1]{\begin{enumerate}[\upshape\bfseries #1]}
\newcommand{\bir}[1][(i)]{\begin{enumerate}[\upshape #1]}
\newcommand{\bic}[1][(i)]{\begin{enumerate}[\upshape\hspace{2\cma}#1]}
\newcommand{\bis}[2][1.]{\begin{enumerate}[\upshape\hspace{#2\parindent}#1]}
\newcommand{\ei}{\end{enumerate}}

\newcommand{\mgp}[1]{\marginpar{\color{red}{#1}}}

% comma is raised when components are quotients

\newcommand\ndots{\raise 0.47ex \hbox {,}\hskip0.06em\cdots %
     \raise 0.47ex \hbox {,}\hskip0.06em} 

%Layout commands


\newcommand{\clearemptydoublepage}{\newpage{\pagestyle{empty}\cleardoublepage}}
\newcommand{\q}{\quad}
\newcommand{\qq}{\qquad}

\newcommand{\vs}[1][3]{\vskip#1pt}
\newcommand{\hs}[1][12]{\hskip#1pt}

\newcommand{\hp}{\hphantom}
\newcommand{\vp}{\vphantom}

\newcommand\cl{\centerline}

\newcommand\nl{\newline}

\newcommand\nd{\noindent}

\newcommand{\nt}{\notag}

% %my private skips; set to 0 to restore default

\newskip\Csmallskipamount                                                
\Csmallskipamount=\smallskipamount
\newskip\Cmedskipamount
\Cmedskipamount=\medskipamount
\newskip\Cbigskipamount
\Cbigskipamount=\bigskipamount

\newcommand\cvs{\vspace\Csmallskipamount}   
\newcommand\cvm{\vspace\Cmedskipamount}
\newcommand\cvb{\vspace\Cbigskipamount}


\newskip\csa
\csa=\smallskipamount

\newskip\cma
\cma=\medskipamount

\newskip\cba
\cba=\bigskipamount

\newdimen\spt
\spt=0.5pt


%%special roster macro

\newcommand\citem{\cvs\advance\itemno by
1{(\romannumeral\the\itemno})\hskip3pt}
\newcommand{\bitem}{\cvm\nd\advance\itemno by
1{\bf\the\itemno}\hspace{\cma}}
\newcommand\cendroster{\cvm\itemno=0}


%New counts

\newcount\itemno
\itemno=0

%Labels

\newcommand{\las}[1]{\label{S:#1}}
\newcommand{\lass}[1]{\label{SS:#1}}
\newcommand{\lae}[1]{\label{E:#1}}
\newcommand{\lat}[1]{\label{T:#1}}
\newcommand{\lal}[1]{\label{L:#1}}
\newcommand{\lad}[1]{\label{D:#1}}
\newcommand{\lac}[1]{\label{C:#1}}
\newcommand{\lan}[1]{\label{N:#1}}
\newcommand{\lap}[1]{\label{P:#1}}
\newcommand{\lar}[1]{\label{R:#1}}
\newcommand{\laa}[1]{\label{A:#1}}
\newcommand{\laas}[1]{\label{Ass:#1}}
\newcommand{\lapp}[1]{\label{Proof:#1}}

%Referencing

\newcommand{\rs}[1]{Section~\ref{S:#1}}
\newcommand{\rss}[1]{Subsection~\ref{SS:#1}}
\newcommand{\rt}[1]{Theorem~\ref{T:#1}}
\newcommand{\rl}[1]{Lemma~\ref{L:#1}}
\newcommand{\rd}[1]{Definition~\ref{D:#1}}
\newcommand{\ras}[1]{Assumption~\ref{Ass:#1}}
\newcommand{\rc}[1]{Corollary~\ref{C:#1}}
\newcommand{\rn}[1]{Number~\ref{N:#1}}
\newcommand{\rp}[1]{Proposition~\ref{P:#1}}
\newcommand{\rr}[1]{Remark~\ref{R:#1}}
\newcommand{\raa}[1]{Axiom~\ref{A:#1}}
\newcommand{\re}[1]{\eqref{E:#1}}
\newcommand{\frc}[1]{Corollary~\ref{C:#1} on page~\tup{\pageref{C:#1}}}
\newcommand{\frt}[1]{Theorem~\ref{T:#1} on page~\tup{\pageref{T:#1}}}
\newcommand{\frl}[1]{Lemma~\ref{L:#1} on page~\tup{\pageref{L:#1}}}
\newcommand{\frr}[1]{Remark~\ref{R:#1} on page~\tup{\pageref{R:#1}}}
\newcommand{\frd}[1]{Definition~\ref{D:#1} on page~\tup{\pageref{D:#1}}}
\newcommand{\fre}[1]{\eqref{E:#1} on page~\tup{\pageref{E:#1}}}
\newcommand{\frss}[1]{Subsection~\ref{SS:#1} on page~\tup{\pageref{SS:#1}}}
\newcommand{\frs}[1]{Section~\ref{S:#1} on page~\tup{\pageref{S:#1}}}
\newcommand{\rpp}[1]{\pageref{Proof:#1}}

%Index
\newcommand{\ind}[1]{#1\index{#1}}

%newtheorems definitions
\newskip\thmskip
\thmskip=\parindent

\newskip\hsk
\setlength{\hsk}{1em/2}%package calc advisable for glue

\newenvironment{hinw}{\labelsep=0pt\begin{list}{}{\labelsep=0pt\itemindent=0pt\labelwidth=0pt\leftmargin=\parindent\rightmargin=0pt\partopsep=\cba}%
\item\it\nopagebreak\nopagebreak}%
{\end{list}}

\newcommand\bh{\begin{hinw}}
\newcommand{\eh}{\end{hinw}}

\newtheoremstyle{normal}% name
  {\cba}%      Space above, empty = `usual value'
  {\cba}%      Space below
  {}% Body font
  {\thmskip}%Indent amount (empty = no indent, \parindent = para indent)
  {\bfseries}% Thm head font
  {.}%        Punctuation after thm head
  {\hsk}%     Space after thm head: " " = normal interword space;
        %       \newline = linebreak
  {}% Thm head spec
%

\newtheoremstyle{abschnitt}% name
  {\cba}%      Space above, empty = `usual value'
  {\cba}%      Space below
  {}% Body font
  {\thmskip}% Indent amount (empty = no indent, \parindent = para indent)
  {\bfseries}% Thm head font
  {.}%        Punctuation after thm head
  {\hsk}%     Space after thm head: " " = normal interword space;
        %       \newline = linebreak
  {}% Thm head spec


\newtheoremstyle{italic}% name
  {\cba}%      Space above, empty = `usual value'
  {\cba}%      Space below
  {\itshape}% Body font
  {\thmskip}%  Indent amount (empty = no indent, \parindent = para indent)
  {\bfseries}% Thm head font
  {.}%        Punctuation after thm head
  {\hsk}%     Space after thm head: " " = normal interword space;
        %       \newline = linebreak
  {}% Thm head spec

\newtheoremstyle{aufgaben}% name
  {\cba}%      Space above, empty = `usual value'
  {\cba}%      Space below
  {}% Body font
  {}%         Indent amount (empty = no indent, \parindent = para indent)
  {\normalsize\bfseries}% Thm head font
  {.}%        Punctuation after thm head
  {\hsk}%     Space after thm head: " " = normal interword space;
        %       \newline = linebreak
  {}% Thm head spec

\newtheoremstyle{break}% name
  {\cba}%      Space above, empty = `usual value'
  {\cba}%      Space below
  {\itshape}% Body font
  {}%         Indent amount (empty = no indent, \parindent = para indent)
  {\bfseries}% Thm head font
  {.}%        Punctuation after thm head
  {\newline}% Space after thm head: \newline = linebreak
  {}%         Thm head spec


%\swapnumbers
\theoremstyle{italic}
\newtheorem{thm}[subsection]{Theorem}
\newtheorem{lem}[subsection]{Lemma}
\newtheorem{prop}[subsection]{Proposition}
\newtheorem{cor}[subsection]{Corollary}

%\swapnumbers
\theoremstyle{normal}
%\newtheorem{abschnitt}[subsection]
\newtheorem{rem}[subsection]{Remark}
\newtheorem{definition}[subsection]{Definition}
\newtheorem{example}[subsection]{Example}
\newtheorem{examples}[subsection]{Examples}
\newtheorem{ex}[subsection]{Exercise}
\newtheorem{note}[subsection]{}
\newtheorem{axiom}[subsection]{Axiom}
\newtheorem{assumption}[subsection]{Assumption}

\theoremstyle{aufgaben}
\newtheorem{exs}[subsection]{Exercises}


%\swapnumbers
\newtheorem{case}{Case}



\numberwithin{equation}{section}
\numberwithin{figure}{section}

%New environments
\newenvironment{textequation}[1][0.8]
{\begin{equation}
\begin{aligned}
\begin{minipage}{#1\linewidth}}
{\end{minipage}
\end{aligned}
\end{equation}
\ignorespacesafterend}

\newcommand{\btext}{\begin{textequation}}
\newcommand{\etext}{\end{textequation}}

\def\hinweis{\@startsection{subsection}{2}%
 \z@{0.7\linespacing\@plus 0.5\linespacing}{0.7\linespacing}%
%%    ^^^   vertical space above   ^^^  ^^  below  ^^^
{\normalfont\itshape\indent}}

\newcounter{hours}\newcounter{minutes}
\newcommand{\printtime}{%
\setcounter{hours}{\time/60}%
\setcounter{minutes}{\time-\value{hours}*60}%
\ifthenelse{\value{minutes}<10}{\thehours :0\theminutes}{\thehours:\theminutes}}

\usepackage[german,english]{babel}
\usepackage{graphicx}
\RequirePackage{amsmath}
\RequirePackage{bm}
\RequirePackage{amssymb}
\RequirePackage{upref}
\RequirePackage{amsthm}
\RequirePackage{enumerate}
\RequirePackage{pb-diagram}
\RequirePackage{amsfonts}
\RequirePackage[mathscr]{eucal}
\RequirePackage{verbatim}
\RequirePackage{xr}
\RequirePackage{graphicx}
%\RequirePackage[pdftex]{floatflt}
\usepackage{calc}
\usepackage{xspace}
%\usepackage{pdfpages}
%\input{microtype}
\usepackage{dsfont}

%\usepackage[parfill]{parskip} 
%\usepackage{microtype}


%\usepackage{epstopdf}
%\DeclareGraphicsRule{.tif}{png}{.png}{`convert #1 `basename #1 .tif`.png}


\makeatletter
\RequirePackage{color}
\newcommand{\ann}[1]{\renewcommand{\@makefnmark}{\mbox{$^{\color{red}{\@thefnmark}}$}}%
\footnote {#1}}
\makeatother

\newcommand{\rmann}{\renewcommand{\ann}[1]{}}

\DeclareMathOperator{\Div}{div}
\newcommand{\fvb}[2]{#1\hspace{0pt}_{\big|_{#2}}}
%\newcommand\chc[3]{{\tilde\varGamma}_{#1#2}^#3}
\newcommand{\sa}{s.a.\@\xspace}




%\Csmallskipamount=0pt 
%\Cmedskipamount=0pt 
%\Cbigskipamount=0pt




%\newcommand{\glqqc}{\tup{\glqq}}
%\newcommand{\grqqc}{\tup{\grqq}}


%\usepackage{fancybox}



%\newcommand{\bis}[2][1.]{\begin{enumerate}[\upshape\hspace{#2\cba}#1]}



%\DeclareMathOperator{\id}{id}
%\DeclareMathOperator{\pr}{pr}
%\newcommand\hra{\hookrightarrow}
%\newcommand\xra{\xrightarrow}
%\newcommand{\Q}{\mathbb Q}
%\RequirePackage{bm}
%\RequirePackage{verbatim}

%\begin{comment}

%\RequirePackage{mathmacrosD}
%\RequirePackage{hyphenation}r
%\RequirePackage{pb-diagram}
\RequirePackage{upref}
\RequirePackage{amsthm}
%\usepackage{amsfonts}
%\usepackage{amsintx}
\RequirePackage{enumerate}%\begin{enumerate}[(i)]
\usepackage[mathscr]{eucal}
%\usepackage{showkeys}

%\usepackage{fancyheadings}
%\usepackage{makeidx}
%\usepackage{lamsarrow,pb-lams}

%\listfiles
%hmargin=1.9cm,bmargin=2.5cm,


%\newcommand{\sqr}{\hbox{\rlap{$\sqcap$}$\sqcup$}}
%\renewcommand{\qed}{\ifmmode\sqr\else{\unskip\nobreak\hfil
%  \penalty50\hskip1em\null\nobreak\hfil\sqr
%  \parfillskip=0pt\finalhyphendemerits=0\endgraf}\fi{}}



%\usepackage[matrix,arrow]{xy}
%\usepackage{pb-diagram}


\usepackage{xr-hyper}
%\usepackage[hyperindex=true, pdfauthor= Claus\  Gerhardt, pdftitle= Analysis\ I, bookmarks=true, extension= pdf, colorlinks=false, plainpages=false,hyperfootnotes=false, debug=false,pagebackref=true]{hyperref}



%\dgHORIZPAD=1em
%\dgVERTPAD=1ex

%\externaldocument{AnalysisII}





%\makeindex

%\nomplabels





%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%

%\newskip\thmskip
%\thmskip=\parindent
%\newtheoremstyle{normal}% name
%  {3pt}%      Space above, empty = `usual value'
%  {3pt}%      Space below
%  {}% Body font
%  {\thmskip}%Indent amount (empty = no indent, \parindent = para indent)
%  {\bfseries}% Thm head font
%  {.}%        Punctuation after thm head
%  {.5em}%     Space after thm head: " " = normal interword space;
%        %       \newline = linebreak
%  {}% Thm head spec
%

%\newtheoremstyle{abschnitt}% name
%  {3pt}%      Space above, empty = `usual value'
%  {3pt}%      Space below
%  {}% Body font
%  {\thmskip}% Indent amount (empty = no indent, \parindent = para indent)
%  {\bfseries}% Thm head font
%  {.}%        Punctuation after thm head
%  {.5em}%     Space after thm head: " " = normal interword space;
%        %       \newline = linebreak
%  {}% Thm head spec

%
%\newtheoremstyle{italic}% name
%  {3pt}%      Space above, empty = `usual value'
%  {3pt}%      Space below
%  {\itshape}% Body font
%  {\thmskip}%  Indent amount (empty = no indent, \parindent = para indent)
%  {\bfseries}% Thm head font
%  {.}%        Punctuation after thm head
%  {.5em}%     Space after thm head: " " = normal interword space;
%        %       \newline = linebreak
%  {}% Thm head spec

%\newtheoremstyle{aufgaben}% name
%  {3pt}%      Space above, empty = `usual value'
%  {3pt}%      Space below
%  {}% Body font
%  {}%         Indent amount (empty = no indent, \parindent = para indent)
%  {\normalsize\bfseries}% Thm head font
%  {.}%        Punctuation after thm head
%  {.5em}%     Space after thm head: " " = normal interword space;
%        %       \newline = linebreak
%  {}% Thm head spec

%

%\swapnumbers
%\theoremstyle{italic}
%\newtheorem{thm}[subsection]{Theorem}
%\newtheorem{lem}[subsection]{Lemma}
%\newtheorem{prop}[subsection]{Proposition}
%\newtheorem{cor}[subsection]{Corollar}

%\swapnumbers
%\theoremstyle{normal}
%\newtheorem{abschnitt}[subsection]
%\newtheorem{rem}[subsection]{Bemerkung}
%\newtheorem{definition}[subsection]{Definition}
%\newtheorem{example}[subsection]{Beispiel}
%\newtheorem{examples}[subsection]{Beispiele}
%\newtheorem{ex}[subsection]{Aufgabe}
%\newtheorem{note}[subsection]{}
%\newtheorem{axiom}[subsection]{Axiom}

%\theoremstyle{aufgaben}
%\newtheorem{exs}[subsection]{Aufgaben}

%
%\swapnumbers
%\newtheorem{case}{Case}

%
%\numberwithin{section}{chapter}
%\numberwithin{equation}{section}
%\numberwithin{figure}{section}

%\renewcommand{\qed}{\mbox{}\hfill\shadowsize1pt\rotatebox{180}{\shadowbox{\hbox{}}}}

%\end{comment}


\listfiles




%\usepackage{pb-diagram}

\setlength{\textwidth}{11.7cm}%Springer
\textheight=18.9cm
%\textheight=17.4cm



%\setlength{\textwidth}{12.7cm}%Springer
%\setlength{\textheight}{20.9cm}

%\newskip\csa
%\csa=\smallskipamount
%
%\newskip\cma
%\cma=\medskipamount
%
%\newskip\cba
%\cba=\bigskipamount
%
%%\newcommand{\nim}{{\hskip2.2ex\not\hskip-1.5ex\im}}
%
%
%\newcommand{\cpl}{\complement}
%\newcommand{\nt}{\notag}



\usepackage{calc}

\newlength{\oddsidemarginlength}
\newlength{\topmarginlength}

\hoffset=0pt
\voffset=0pt

 \textwidth = 11.7cm%your choice (any dimension is allowed)
\setlength{\textheight}{19cm -\topskip}

\newcounter{numberoflines}
\newcounter{tempcc}
\setcounter{numberoflines}{\textheight}
\setcounter{tempcc}{\baselineskip}
\setcounter{numberoflines}{\value{numberoflines}/\value{tempcc}}
\setlength{\textheight}{\baselineskip*\value{numberoflines}+\topskip}

\setlength{\oddsidemarginlength}{(\paperwidth-\textwidth)/2-1in}
\setlength{\topmarginlength}{(\paperheight-\textheight)/2-1in}

\oddsidemargin=\oddsidemarginlength
\evensidemargin=\oddsidemargin
\topmargin=\topmarginlength
\headsep=20pt







%\baselineskip=12pt

%\parindent=18pt
%\normalparindent=\thmskip

%\parskip=0pt plus 1pt

\hfuzz=1pt

%\frenchspacing



\tolerance=500

%\abovedisplayskip=3mm plus6pt minus 4pt

%\belowdisplayskip=3mm plus6pt minus 4pt

%\abovedisplayshortskip=0mm plus6pt minus 2pt

%\belowdisplayshortskip=2mm plus4pt minus 4pt


%\predisplaypenalty=0

%\clubpenalty=10000

%\widowpenalty=10000

%\setlength{\abovedisplayskip}{\cma}
%\setlength{\belowdisplayskip}{\cma}



%\end{comment}
%\usepackage[verbose,papersize={15.5cm,22cm}, pdftex]{geometry} 
%\geometry{twosideshift=0pt, textheight=17.4cm, textwidth= 11.9cm, bmargin=2.5cm, lmargin=1.9cm, rmargin=1.9cm}


%\hyphenation{Pro-po-si-tion Lem-ma Theo-rem Be-mer-kung geo-me-trisch Ses-qui-li-near-form  Rau-mes Zu-sam-men-hangs-kom-po-nen-te Kom-po-nen-ten Dif-fe-ren-tial-glei-chun-gen Ex-po-nen-tial-funk-tion da-rauf Ska-lar-pro-dukt-raum Kro-necker-sym-bol Dif-fe-ren-tial-rech-nung Mi-ni-mal-fol-ge Cau-chy-fol-ge be-in-hal-ten Kon-ver-genz-kri-te-ri-um Schwarz-sche Zu-sam-men-hangs-kom-po-nen-ten Zei-chen Bild-raum be-trach-ten Po-tenz-rei-he Po-ly-zy-lin-der ei-ner sinn-vol-len}



%\input{HyphenationD}

%\includeonly{schmutztitelI , Chap0fin,Chap1fin, Chap2in}%, Chap3in}%, Chap4in}
%\includeonly{schmutztitelI , titleI , impressumI,  prefaceI}


%\includeonly{schmutztitelI , titleI , impressumI,  preface, Chap0fin, Chap1fin}



%\opengraphsfile{aIpic}

%\setlength{\fboxsep}{0pt}

%\headshape{1}{1}{true}


\usepackage[colorlinks=true,linkcolor=blue,citecolor=blue,urlcolor=blue]{hyperref}  

\begin{document}
\begin{thebibliography}{10}

\bibitem{adm:old}
R.~Arnowitt, S.~Deser, and C.~W. Misner, \emph{The dynamics of general
  relativity}, Gravitation: an introduction to current research (Louis Witten,
  ed.), John Wiley, New York, 1962, pp.~227--265.

\bibitem{baer:book}
Christian Baer, Nicolas Ginoux, and Frank Pfaeffle, \emph{{Wave Equations on
  Lorentzian Manifolds and Quantization}}, 2008,
  {\href{http://arXiv.org/abs/0806.1036}{arXiv:0806.1036}}.

\bibitem{robinson:book2}
Ola Bratteli and Derek~W. Robinson, \emph{Operator algebras and quantum
  statistical mechanics. 2}, second ed., Texts and Monographs in Physics,
  Springer-Verlag, Berlin, 1997, Equilibrium states. Models in quantum
  statistical mechanics.

\bibitem{friedlander:book}
F.~G. Friedlander, \emph{The wave equation on a curved space-time}, Cambridge
  University Press, Cambridge-New York-Melbourne, 1975, Cambridge Monographs on
  Mathematical Physics, No. 2.

\bibitem{cg:cp}
Claus Gerhardt, \emph{Curvature {P}roblems}, Series in Geometry and Topology,
  vol.~39, International Press, Somerville, MA, 2006.

\bibitem{cg:qfriedman}
\bysame, \emph{{Quantum cosmological Friedman models with an initial
  singularity}}, Class. Quantum Grav. \textbf{26} (2009), no.~1, 015001,
  {\href{http://arXiv.org/abs/0806.1769}{arXiv:0806.1769}},
  {\href{http://dx.doi.org/10.1088/0264-9381/26/1/015001}{doi:10.1088/0264-9381/26/1/015001}}.

\bibitem{cg:pdeII}
\bysame, \emph{{Partial differential equations II}}, Lecture Notes, University
  of Heidelberg, 2013,
  {\href{http://www.math.uni-heidelberg.de/studinfo/gerhardt/PDE2.pdf}{pdf
  file}}.

\bibitem{cg:qgravity}
\bysame, \emph{{The quantization of gravity in globally hyperbolic
  spacetimes}}, Adv. Theor. Math. Phys. \textbf{17} (2013), no.~6, 1357--1391,
  {\href{http://arXiv.org/abs/1205.1427}{arXiv:1205.1427}},
  {\href{http://dx.doi.org/10.4310/ATMP.2013.v17.n6.a5}{doi:10.4310/ATMP.2013.v17.n6.a5}}.

\bibitem{cg:qbh}
\bysame, \emph{{The quantization of a black hole}},  (2016),
  {\href{http://arXiv.org/abs/1608.08209}{arXiv:1608.08209}}.

\bibitem{cg:uf4}
\bysame, \emph{{The quantum development of an asymptotically Euclidean Cauchy
  hypersurface}},  (2016),
  {\href{http://arXiv.org/abs/1612.03469}{arXiv:1612.03469}}.

\bibitem{cg:qbh2}
\bysame, \emph{{The quantization of a Kerr-AdS black hole}}, Advances in
  Mathematical Physics \textbf{vol. 2018} (2018), Article ID 4328312, 10 pages,
  {\href{http://arXiv.org/abs/1708.04611}{arXiv:1708.04611}},
  {\href{http://dx.doi.org/10.1155/2018/4328312}{doi:10.1155/2018/4328312}}.

\bibitem{ginoux:fredenhagen}
Nicolas Ginoux, \emph{Linear wave equations}, Quantum field theory on curved
  spacetimes, Lecture Notes in Phys., vol. 786, Springer, Berlin, 2009,
  {\href{http://dx.doi.org/10.1007/978-3-642-02780-2\_3}{doi:10.1007/978-3-642-02780-2\_3}},
  pp.~59--84.

\bibitem{gunther:book}
Paul G{\"u}nther, \emph{Huygens' principle and hyperbolic equations},
  Perspectives in Mathematics, vol.~5, Academic Press, Inc., Boston, MA, 1988,
  With appendices by V. W{\"u}nsch.

\bibitem{hamilton:ricci}
Richard~S. Hamilton, \emph{Three-manifolds with positive {R}icci curvature}, J.
  Differential Geom. \textbf{17} (1982), no.~2, 255--306.

\end{thebibliography}



\flushbottom





%\larger[1]
%\frontmatter

\title{The quantization of gravity}

% author one information
\author{Claus Gerhardt}
\address{Ruprecht-Karls-Universit\"at, Institut f\"ur Angewandte Mathematik,
Im Neuenheimer Feld 205, 69120 Heidelberg, Germany}
%\curraddr{}
\email{\href{mailto:gerhardt@math.uni-heidelberg.de}{gerhardt@math.uni-heidelberg.de}}
\urladdr{\href{http://www.math.uni-heidelberg.de/studinfo/gerhardt/}{http://www.math.uni-heidelberg.de/studinfo/gerhardt/}}
%\thanks{This work was supported by the DFG}


% author two information
%\author{}
%\address{}
%\curraddr{}
%\email{}
%\thanks{}
%
\subjclass[2000]{83,83C,83C45}
\keywords{unified field theory, quantization of gravity, quantum gravity, gravitational waves, graviton}
\date{\today}
%
% at present the "communicated by" line appears only in ERA and PROC
%\commby{}

%\dedicatory{}


\begin{abstract} 
In a former paper we proposed a model for the quantization of gravity by working in a bundle $E$ where we  realized the Hamilton constraint as the Wheeler-DeWitt equation. However, the corresponding operator only acts in the fibers and not in the base space. Therefore, we now discard the Wheeler-DeWitt equation and express the Hamilton constraint differently, either with the help of the Hamilton equations or by employing a geometric evolution equation. There are two possible modifications possible which both are equivalent to the Hamilton constraint and which lead to two new models. In the first model we obtain a hyperbolic operator that acts in the fibers as well as in the base space and we can construct a symplectic vector space and a Weyl system.

\nd
In the second model  the resulting equation is a wave equation in $\so \times (0,\infty)$ valid in points $(x,t,\xi)$ in $E$ and we look for solutions for each fixed $\xi$. This set of equations contains as a special case the  equation of a  quantized cosmological Friedmann universe without matter but with a cosmological constant, when we look for solutions which only depend on $t$. Moreover, in case $\so$ is compact we prove a spectral resolution of the equation.
\end{abstract}

%\include{schmutztitelI}
%\include{titleI}
%\include{impressumI}
%\include{preface}
\maketitle


\tableofcontents


\setcounter{section}{0}
\section{Introduction}

The quantization of gravity is hampered by the fact that the Einstein-Hilbert Lagrangian is singular. Switching to a Hamiltonian setting requires to impose two constraints, the Hamilton constraint and the diffeomorphism constraint. Though we were able to eliminate the diffeomorphism constraint in a recent paper \cite{cg:qgravity}, the Hamilton constraint is a serious obstacle. Quantization of a Hamiltonian setting requires a model in which the quantized variables, which turn into operators, act, and, in case of constraints, preferably given as an equation, to quantize this equation.

In the former paper we proposed a quantization of gravity by working in a fiber bundle $E$ with base space $\so$ after quantization, the Hamilton function $H$ was transformed to an hyperbolic operator $\hat H$ and the Hamilton condition, which could be expressed by
\begin{equation}\lae{1.1}
H=0,
\end{equation}
was transformed to the Wheeler-DeWitt equation
\begin{equation}
\hat H u=0
\end{equation}
in the bundle $E$. However, the operator $\hat H$ acts only in the fibers, there is no differentiation in the base space $\so$, though the solutions are defined in $E$. This seems to be unsatisfactory.

In this paper we want to offer a better quantization model: We are still working in the bundle $E$, but we discard the Wheeler-DeWitt equation, i.e., we do not express the Hamilton constraint by equation \re{1.1} but differently using the Hamilton equations. The second Hamilton equation has the form
\begin{equation}
\dot \pi^{ij}=-\dde{\mc H}{g_{ij}},
\end{equation}
or equivalently,
\begin{equation}
\dot\pi^{ij}=\{\pi^{ij},\mc H\},
\end{equation}
where we use a Hamiltonian density at the moment. Hence we have the identity
\begin{equation}\lae{1.5}
g_{ij}\{\pi^{ij},\mc H\}=-g_{ij}\dde{\mc H}{g_{ij}}
\end{equation}
which is a scalar equation. 

The Hamilton constraint can be expressed in the form
\begin{equation}
\abs A^2-H^2=(R-2\Lam).
\end{equation}
Looking at the right-hand side of \re{1.5} the term $\abs A^2-H^2$, which will be transformed to be the main part of the hyperbolic operator, occurs on the right-hand side in two places. Replacing $\abs A^2-H^2$ on the right side by $(R-2\Lam)$ will give an equation that defines the Hamilton constraint.

We developed two models: In the first model we replaced $\abs A^2-H^2$ partially in \re{1.5}. The quantization of the modified equation  then leads to a hyperbolic equation
\begin{equation}
Pu=0
\end{equation}
in $E$, where $P$ acts in the fibers as well as in $\so$. $P$ is a symmetric operator and with the help of its Green's operator one can define a symplectic vector space and then a Weyl system, or a quantum field.

In the second model we use a geometric evolution equation to express the Hamilton constraint  by replacing $\abs A^2-H^2$ completely in the evolution equation. After quantization we then obtain a wave equation in $E$
\begin{equation}\lae{1.8}
\frac1{32}\frac{n^2}{n-1}\Ddot u
-(n-1) t^{2-\frac4n}\D u-\frac {n}2t^2(t^{-\frac4n}R-2\Lam)u=0
\end{equation}
in points $(x,t,\xi)\in E$, where a metric $g_{ij}$ in the fiber over $x\in\so$ has the form
\begin{equation}
g_{ij}=t^\frac4n\s_{ij}(x,\xi)
\end{equation}
and the Laplacian in \re{1.8} is defined with respect to $\s_{ij}$. Hence, for any $\xi$ we have a wave equation in
\begin{equation}
\so\times \R[*]_+
\end{equation}
with solutions $u=u(x,t,\xi)$. We prove that solutions of the corresponding Cauchy problems exist and are smooth in all variables.

This second model seems to be the right model since it contains the quantization of a cosmological Friedmann universe, without matter but with a cosmological constant, as a special case by choosing $\s_{ij}$ to be the metric of a space of constant curvature and by assuming $u=u(t)$. Equation \re{1.8} is in this case identical to the quantized Friedmann equation up to the last constant.

Moreover, assuming $\so$ to be compact we also prove a spectral resolution of equation \re{1.8}, by constructing a countable basis of solutions of the form
\begin{equation}
u=w(t)v(x),
\end{equation}
where $v$ is an eigenfunction of the problem
\begin{equation}\lae{1.12}
-(n-1)\D v-\frac{n}2R v=\mu v
\end{equation}
in $\so$ with $\mu>0$  and $w$ an eigenfunction of an ODE. These solutions have finite energy, \cf\fre{6.49}.

The results for the first model are proved and described in detail in \rs{4} and \rs {5}. The results for the second model are proved in \rs{6}. Here is a more formal summary of the results of the second model:
\bt
Let $(\so,\s_{ij})$ be a given connected, smooth and complete  $n$-dimensional Riemannian mani\-fold   and let 
\begin{equation}
Q=\so\times \R[*]_+
\end{equation}
be the corresponding globally hyperbolic spacetime equipped with the Lorentzian metric \re{6.17} or, if necessary, with \re{6.18}, then the hyperbolic equation
\begin{equation}
\frac1{32}\frac{n^2}{n-1}\Ddot u
-(n-1) t^{2-\frac4n}\D u-\frac {n}2 t^{2-\frac4n}Ru+nt^2\Lam u=0,
\end{equation}
where the Laplacian and the scalar curvature correspond to the metric $\s_{ij}$, describes a model for quantum gravity. If  $\so$ is compact a spectral resolution of this equation has been proved in the theorem below.
\et

\bt
Assume $n\ge 2$ and  $\so$ to be compact and let $(v,\mu)$ be a solution of the eigenvalue problem \re{1.12} with $\mu>0$, then there exist countably many solutions $(w_i,\Lam_i)$ of the implicit eigenvalue problem \re{6.33} such that
\begin{equation}
\Lam_i<\Lam_{i+1}<\cdots <0,
\end{equation}
\begin{equation}
\lim_i\Lam_i=0,
\end{equation}
and such that the functions
\begin{equation}
u_i=w_i v
\end{equation}
are solutions of the wave equations \re{1.8}. The transformed eigenfunctions
\begin{equation}
\tilde w_i(t)=w_i(\lam_i^{\frac n{4(n-1)}}t), 
\end{equation}
where
\begin{equation}
\lam_i=(-\Lam_i)^{-\frac{n-1}n}
\end{equation}
form a basis of the corresponding Hilbert space $H$ and also of $L^2(\R[*]_+,\Cc)$.
\et




\section{Definitions and notations}
The main objective of this section is to state the equations of Gau{\ss}, Codazzi,
and Weingarten for spacelike hypersurfaces $M$ in a \di {(n+1)} Lorentzian
manifold
$N$.  Geometric quantities in $N$ will be denoted by
$(\bar g_{ \al \bet}),(\riema  \al \bet \ga \de)$, etc., and those in $M$ by $(g_{ij}), 
(\riem ijkl)$, etc.. Greek indices range from $0$ to $n$ and Latin from $1$ to $n$;
the summation convention is always used. Generic coordinate systems in $N$ resp.
$M$ will be denoted by $(x^ \al)$ \resp $(\x^i)$. Covariant differentiation will
simply be indicated by indices, only in case of possible ambiguity they will be
preceded by a semicolon, i.e., for a function $u$ in $N$, $(u_ \al)$ will be the
gradient and
$(u_{ \al \bet})$ the Hessian, but e.g., the covariant derivative of the curvature
tensor will be abbreviated by $\riema  \al \bet \ga{ \de;\e}$. We also point out that
\begin{equation}
\riema  \al \bet \ga{ \de;i}=\riema  \al \bet \ga{ \de;\e}x_i^\e
\end{equation}
with obvious generalizations to other quantities.

Let $M$ be a \tit{spacelike} hypersurface, i.e., the induced metric is Riemannian,
with a differentiable normal $\n$ which is timelike.

In local coordinates, $(x^ \al)$ and $(\x^i)$, the geometric quantities of the
spacelike hypersurface $M$ are connected through the following equations
\begin{equation}\lae{01.2}
x_{ij}^ \al= h_{ij}\n^ \al
\end{equation}
the so-called \tit{Gau{\ss} formula}. Here, and also in the sequel, a covariant
derivative is always a \tit{full} tensor, i.e.

\begin{equation}
x_{ij}^ \al=x_{,ij}^ \al-\ch ijk x_k^ \al+ \cha  \bet \ga \al x_i^ \bet x_j^ \ga.
\end{equation}
The comma indicates ordinary partial derivatives.

In this implicit definition the \tit{second fundamental form} $(h_{ij})$ is taken
with respect to $\n$.

The second equation is the \tit{Weingarten equation}
\begin{equation}
\n_i^ \al=h_i^k x_k^ \al,
\end{equation}
where we remember that $\n_i^ \al$ is a full tensor.

Finally, we have the \tit{Codazzi equation}
\begin{equation}
h_{ij;k}-h_{ik;j}=\riema \al \bet \ga \de\n^ \al x_i^ \bet x_j^ \ga x_k^ \de
\end{equation}
and the \tit{Gau{\ss} equation}
\begin{equation}
\riem ijkl=- \{h_{ik}h_{jl}-h_{il}h_{jk}\} + \riema  \al \bet\ga \de x_i^ \al x_j^ \bet
x_k^ \ga x_l^ \de.
\end{equation}

Now, let us assume that $N$ is a globally hyperbolic Lorentzian manifold with a
 Cauchy surface. 
$N$ is then a topological product $I\times \mc S_0$, where $I$ is an open interval,
$\mc S_0$ is a  Riemannian manifold, and there exists a Gaussian coordinate
system
$(x^ \al)$, such that the metric in $N$ has the form 
\begin{equation}\lae{01.7}
d\bar s_N^2=e^{2\psi}\{-{dx^0}^2+\s_{ij}(x^0,x)dx^idx^j\},
\end{equation}
where $\s_{ij}$ is a Riemannian metric, $\psi$ a function on $N$, and $x$ an
abbreviation for the spacelike components $(x^i)$. 
We also assume that
the coordinate system is \tit{future oriented}, i.e., the time coordinate $x^0$
increases on future directed curves. Hence, the \tit{contravariant} timelike
vector $(\x^ \al)=(1,0,\dotsc,0)$ is future directed as is its \tit{covariant} version
$(\x_ \al)=e^{2\psi}(-1,0,\dotsc,0)$.

Let $M=\graph \fv u\so$ be a spacelike hypersurface
\begin{equation}
M=\set{(x^0,x)}{x^0=u(x),\,x\in\mc S_0},
\end{equation}
then the induced metric has the form
\begin{equation}
g_{ij}=e^{2\psi}\{-u_iu_j+\s_{ij}\}
\end{equation}
where $\s_{ij}$ is evaluated at $(u,x)$, and its inverse $(g^{ij})=(g_{ij})^{-1}$ can
be expressed as
\begin{equation}\lae{01.10}
g^{ij}=e^{-2\psi}\{\s^{ij}+\frac{u^i}{v}\frac{u^j}{v}\},
\end{equation}
where $(\s^{ij})=(\s_{ij})^{-1}$ and
\begin{equation}\lae{01.11}
\begin{aligned}
u^i&=\s^{ij}u_j\\
v^2&=1-\s^{ij}u_iu_j\equiv 1-\abs{Du}^2.
\end{aligned}
\end{equation}
Hence, $\graph u$ is spacelike if and only if $\abs{Du}<1$.

The covariant form of a normal vector of a graph looks like
\begin{equation}
(\n_ \al)=\pm v^{-1}e^{\psi}(1, -u_i).
\end{equation}
and the contravariant version is
\begin{equation}
(\n^ \al)=\mp v^{-1}e^{-\psi}(1, u^i).
\end{equation}
Thus, we have
\br Let $M$ be spacelike graph in a future oriented coordinate system. Then the
contravariant future directed normal vector has the form
\begin{equation}
(\n^ \al)=v^{-1}e^{-\psi}(1, u^i)
\end{equation}
and the past directed
\begin{equation}\lae{01.15}
(\n^ \al)=-v^{-1}e^{-\psi}(1, u^i).
\end{equation}
\er

In the Gau{\ss} formula \re{01.2} we are free to choose the future or past directed
normal, but we stipulate that we always use the past directed normal.
Look at the component $ \al=0$ in \re{01.2} and obtain in view of \re{01.15}

\begin{equation}\lae{01.16}
e^{-\psi}v^{-1}h_{ij}=-u_{ij}- \cha 000\mspace{1mu}u_iu_j- \cha 0j0
\mspace{1mu}u_i- \cha 0i0\mspace{1mu}u_j- \cha ij0.
\end{equation}
Here, the covariant derivatives are taken with respect to the induced metric of
$M$, and
\begin{equation}
-\cha ij0=e^{-\psi}\bar h_{ij},
\end{equation}
where $(\bar h_{ij})$ is the second fundamental form of the hypersurfaces
$\{x^0=\const\}$.

An easy calculation shows
\begin{equation}
\bar h_{ij}e^{-\psi}=-\tfrac{1}{2}\dot\s_{ij} -\dot\psi\s_{ij},
\end{equation}
where the dot indicates differentiation with respect to $x^0$.

\section{Combining the Hamilton equations with the Hamilton constraint}\las{3}

Let $N=N^{n+1}$ be a globally hyperbolic spacetime with metric $\bar g_{\al\bet}$. We consider the Einstein-Hilbert functional
\begin{equation}\lae{3.1}
J=\int_N(\bar R-2\Lam)
\end{equation}
with cosmological constant $\Lam$ and want to write it in a form such that the Lagrangian density is regular with respect to the variables $g_{ij}$ so that we can switch to an equivalent Hamiltonian setting for these components. Let $x^0$ be time function that will split the metric such that the metric can be expressed in the form
\begin{equation}\lae{3.2}
d\bar s^2=-w^2 (dx^0)^2+g_{ij}dx^idx^j,
\end{equation}
where $(x^i)$ are local coordinates of a coordinate slice
\begin{equation}
\so=\{x^0=\const\}
\end{equation}
and 
\begin{equation}
0<w\in C^\un(N).
\end{equation}
Let us define the level sets
\begin{equation}
M(t)=\{x^0=t\}
\end{equation}
and, assuming  $0\in x^0(N)$, set
\begin{equation}
\so=M(0).
\end{equation}
The coordinate system should also be future oriented such that $\{x^0>0\}$ is the future development of $\so$.

Let $h_{ij}$ be the second fundamental form of the slices $M(t)$ with respect to the past directed normal, i.e., the Gaussian formula looks like
\begin{equation}
x^\al_{ij}=h_{ij}\nu,
\end{equation}
where $\nu$ is the past directed normal. Then
\begin{equation}\lae{3.8}
h_{ij}=-\tfrac12 \dot g_{ij}w^{-1}
\end{equation}
and the functional \re{3.1} can be expressed in the form
\begin{equation}\lae{3.9}
J=\int_a^b\int_\Om\{\abs A^2-H^2+(R-2\Lam)\}w\sqrt g,
\end{equation}
where $\Om\su N$ is some open subset of $\R[n]$, $R$ the scalar curvature of $M(t)$,
\begin{equation}
H=g^{ij}h_{ij}
\end{equation}
 the mean curvature and
\begin{equation}
\abs A^2=h_{ij}h^{ij},
\end{equation}
\cf \cite[equ. (3.37)]{cg:qgravity}. This way of expressing the Einstein-Hilbert functional is known as the ADM approach, see \cite{adm:old}.

Let $F=F(h_{ij})$ be the scalar curvature operator
\begin{equation}
F=\tfrac12(H^2-\abs A^2)
\end{equation}
and let
\begin{equation}
F^{ij,kl}=g^{ij}g^{kl}-\tfrac12\{g^{ik}g^{jl}+g^{il}g^{jk}\}
\end{equation}
be its Hessian, then
\begin{equation}\lae{3.14}
F^{ij,kl}h_{ij}h_{kl}=2F=H^2-\abs A^2
\end{equation}
and
\begin{equation}\lae{3.15}
F^{ij}=F^{ij,kl}h_{kl}=Hg^{ij}-h^{ij}.
\end{equation}
In physics
\begin{equation}\lae{3.16}
G^{ij,kl}=-F^{ij,kl}
\end{equation}
is known as the DeWitt metric.

Combining \re{3.8} and \re{3.14} $J$ can be expressed in the form
\begin{equation}\lae{3.17}
J=\int_a^b\int_\Om\{\tfrac14 G^{ij,kl}\dot g_{ij}\dot g_{kl}w^{-2}+(R-2\Lam)\}w\sqrt g.
\end{equation}
The Lagrangian density $\mc L$ is a regular Lagrangian with respect to the variables $g_{ij}$. Define the conjugate momenta
\begin{equation}\lae{3.18}
\begin{aligned}
\pi^{ij}=\frac{\pa\mc L}{\pa g_{ij}}&=\tfrac12 G^{ij,kl}\dot g_{kl}w^{-1}\sqrt g\\
&= -G^{ij,kl}h_{kl}\sqrt g
\end{aligned}
\end{equation}
and the Hamiltonian density
\begin{equation}
\begin{aligned}
\mc H&=\pi^{ij}\dot g_{ij}-\mc L\\
&=\frac1{\sqrt g}wG_{ij,kl}\pi^{ij}\pi^{kl}-(R-2\Lam)w\sqrt g,
\end{aligned}
\end{equation}
where
\begin{equation}
G_{ij,kl}=\tfrac12\{g_{ik}g_{jk}+g_{il}g_{jk}\}-\tfrac1{n-1}g_{ij}g_{kl}
\end{equation}
is the inverse of $G^{ij,kl}$.

Let us now consider an arbitrary variation of $g_{ij}$ with compact support
\begin{equation}
g_{ij}(\e)=g_{ij}+\e\om_{ij},
\end{equation}
where $\om_{ij}=\om_{ij}(t,x)$ is an arbitrary smooth, symmetric tensor with compact support in $\Om$. The vanishing of the first variation leads to the Euler-Lagrange equations
\begin{equation}\lae{3.22}
G_{ij}+\Lam g_{ij}=0,
\end{equation}
i.e., to the tangential Einstein equations. We obtain these equations by either varying \re{3.1} or \re{3.9}.

To obtain the full Einstein equations we impose the Hamilton constraint, namely, that the Hamiltonian density vanishes, or equivalently, that the normal component of the Einstein equations is satisfied
\begin{equation}\lae{3.23}
G_{\al\bet}\nu^\al\nu^\bet-\Lam=0.
\end{equation}
We then conclude that any metric $(\bar g_{\al\bet})$ satisfying \re{3.2}, \re{3.22} as well as \re{3.23} has the property that it is a stationary point for the functional \re{3.1} in the class of metrics which can be split according to \re{3.2}. Applying then a former result \cite[Theorem 3.2]{cg:qgravity} we deduce that $\bar g_{\al\bet}$ satisfies the full Einstein equations.

The Lagrangian density $\mc L$ in \re{3.17} is regular with respect to the variables $g_{ij}$, hence the tangential Einstein equations are equivalent to the Hamilton equations
\begin{equation}\lae{3.24}
\dot g_{ij}=\frac{\de \mc H}{\de \pi^{ij}} 
\end{equation}
and
\begin{equation}\lae{3.40}
\dot\pi^{ij}=-\dde{\mc H}{g_{ij}},
\end{equation}
where the differentials on the right-hand side of these equations are variational or functional derivatives, i.e., they are the Euler-Lagrange operators of the corresponding functionals with respect to the indicated variables, in this case, the functional is
\begin{equation}
\int_\Om\mc H,
\end{equation}
where $\so$ is locally parameterized over $\Om\su\R[n]$. Occasionally we shall also write
\begin{equation}
\int_\so\mc H
\end{equation}
by considering $\so$ simply to be a parameter domain without any intrinsic volume element.

We have therefore proved:
\bt\lat{3.1}
Let $N=N^{n+1}$ be a globally hyperbolic spacetime and let the metric $\bar g_{\al\bet}$ be expressed as in \re{3.2}. Then, the metric satisfies the full Einstein equations if and only if the metric is a solution of the  Hamilton equations \re{3.24} and \re{3.40} and of the equation \re{3.23} which is equivalent to 
\begin{equation}
\mc H=0
\end{equation}
and is called the Hamiltonian constraint. These equations are equations for the variables $g_{ij}$. The function $w$ is merely part of the equations and not looked at as a variable though it is of course specified in the component $\bar g_{00}$.
\et

We define the Poisson brackets 
\begin{equation}
\{u,v\}=\dde u{g_{kl}}\dde v{\pi^{kl}}-\dde u{\pi^{kl}}\dde v{g_{kl}}
\end{equation}
and obtain
\begin{equation}\lae{4.2.1}
\{g_{ij},\pi^{kl}\}=\de^{kl}_{ij},
\end{equation}
where
\begin{equation}
\de^{kl}_{ij}=\tfrac12\{\de^k_i\de^l_j+\de^l_i\de^k_j\}.
\end{equation}
Then, the second Hamilton equation can also be expressed as
\begin{equation}\lae{3.32}
\dot \pi^{ij}=\{\pi^{ij},\mc H\}.
\end{equation}
In the next section we want to quantize this Hamiltonian setting and especially the Hamiltonian constraint. In order to achieve this we shall express the equation \re{3.40}, \re{3.24} and \re{3.23} by a set of equivalent equations, namely, \re{3.40}, \re{3.24} and \re{3.33} 
\begin{equation}\lae{3.33}
\begin{aligned}
g_{ij}\{\pi^{ij},\mc H\}&=(n-1) (R-2\Lam)w\sqrt g-Rw\sqrt g-(n-1)\tilde\D w \sqrt g \\
&\q\qq -\frac1{\sqrt g}G_{rs,kl}\pi^{rs}\pi^{kl}w,
\end{aligned}
\end{equation}
where $\tilde \D$ is the Laplacian with respect to the metric $g_{ij}$.
Let us formulate this claim as a theorem:
\bt
Let $N=N^{n+1}$ be a globally hyperbolic spacetime and let the metric $\bar g_{\al\bet}$ be expressed as in \re{3.2}. Then, the metric satisfies the full Einstein equations if and only if the metric is a solution of the  Hamilton equations \re{3.24} and \re{3.40} and of the equation \re{3.33}.
\et
\bp
The second Hamilton equation states
\begin{equation}
\dot\pi^{ij}=-\frac{\de\mc H}{\de g_{ij}},
\end{equation}
which is of course equal to \re{3.32}, and 
\begin{equation}\lae{4.10.1}
\begin{aligned}
-\dde{\mc H}{g_{ij}}=-\frac{\pa}{\pa g_{ij}}(\frac1{\sqrt g}G_{rs,kl}\pi^{rs}\pi^{kl})w+\dde{((R-2\Lam)w\sqrt g)}{g_{ij}}.
\end{aligned}
\end{equation}
In the lemma below we shall prove
\begin{equation}\lae{4.11.1}
\begin{aligned}
\dde{((R-2\Lam)w\sqrt g)}{g_{ij}}&=\tfrac12Rg^{ij}w\sqrt g-R^{ij}w\sqrt g\\
&\q+\{w^{ij}-\D w g^{ij}-\Lam g^{ij}w\}\sqrt g
\end{aligned}
\end{equation}
and a simple but somewhat lengthy computation will reveal
\begin{equation}\lae{4.12.1}
\begin{aligned}
-\frac{\pa}{\pa g_{ij}}(\frac1{\sqrt g}G_{rs,kl}\pi^{rs}\pi^{kl})w&= \tfrac12(\abs A^2-H^2)g^{ij}w\sqrt g\\
&\msp[-100]-2\pi^i_r\pi^{rj}w\frac1{\sqrt g} +\frac2{n-1}\pi^{ij}\pi^r_rw\frac1{\sqrt g},
\end{aligned}
\end{equation}
where the indices are lowered with the help of $g_{ij}$ and we further conclude
\begin{equation}\lae{4.12.1.1}
\begin{aligned}
\msp[-79]&-g_{ij}\frac{\pa}{\pa g_{ij}}(\frac1{\sqrt g}G_{rs,kl}\pi^{rs}\pi^{kl})w
\\&= \frac n2(\abs A^2-H^2)w\sqrt g
-2(\abs A^2-H^2)w{\sqrt g}\\
&=(\frac n2-1)(\abs A^2-H^2)w\sqrt g-\frac1{\sqrt g}G_{rs,kl}\pi^{rs}\pi^{kl}w
\end{aligned}
\end{equation}


On the other hand, the Hamilton density is equal to 
\begin{equation}\lae{4.8.1}
\mc H=-2\{\ein-\Lam\}w\sqrt g
\end{equation}
because of the Gau{\ss} equation. Hence,
\begin{equation}\lae{3.39} 
\frac12\{\abs A^2-H^2\}w\sqrt g=\frac12 (R-2\Lam)w\sqrt g
\end{equation}
iff the Hamilton constraint is valid, from which the proof of the theorem immediately follows.
\ep

\bl\lal{3.3}
Let $M$ be a Riemannian manifold with metric $g_{ij}$, scalar curvature $R$ and let $w\in C^2(M)$ and $\Lam\in\R[]$, then the equation \re{4.11.1} is valid.
\el

\bp
It suffices to consider the term 
\begin{equation}
\dde{(Rw\sqrt g)}{g_{ij}},
\end{equation}
since the result for the second term is trivial.

Let $\Om\su M$ be open and bounded and define the functional
\begin{equation}
J=\int_\Om Rw\sqrt g.
\end{equation}
Let $g_{ij}(\e)$ be a variation of $g_{ij}$ with support in $\Om$ such that
\begin{equation}
g_{ij}=g_{ij}(0)
\end{equation}
and denote differentiation with respect to $\e$ by a dot or prime, then the first variation of $J$ with respect to this variation is equal to
\begin{equation}
\begin{aligned}
\dot J(0)&=\int_\Om\{\dot g^{ij}R_{ij}+g^{ij}\dot R_{ij}\}w\sqrt g+\int_\Om Rw\sqrt g'.
\end{aligned}
\end{equation}
Again we only consider the non-trivial term
\begin{equation}\lae{4.19.1}
\int_\Om g^{ij}\dot R_{ij}w\sqrt g.
\end{equation}
It is well known that
\begin{equation}
\dot R_{ij}=-(\dot\C^k_{ik})_{;j}+(\dot\C^k_{ij})_{;k},
\end{equation}
where the semicolon indicates covariant differentiation, $\dot\C^k_{ij}$ is a tensor. Hence, we deduce that \re{4.19.1} is equal to
\begin{equation}
\int_\Om\{g^{ij}\dot\C^k_{ik}w_j-g^{ij}\dot\C^k_{ij}w_k\}\sqrt g
\end{equation}
which in turn can be expressed as
\begin{equation}
\begin{aligned}
&\int_\Om g^{ij}g^{kl}\tfrac12(\dot g_{il;k}+\dot g_{kl;i}-\dot g_{ik;l})w_j\\
&-\int_\Om g^{ij}g^{kl}\tfrac12(\dot g_{il;j}+\dot g_{jl;i}-\dot g_{ij;l})w_k,
\end{aligned}
\end{equation}
where we omitted the notation of the density $\sqrt g$. Let us agree that each row of the preceding expression contains three integrals. Then the first integrals in each row cancel each other, the second in the first row is equal to the third integral in the second row and the third integral in the first row is equal to the second integral in the second row. Therefore, we obtain by integrating by parts
\begin{equation}
\begin{aligned}
-\int_\Om\D wg^{kl}\dot g_{kl}+\int_\Om w^l_i\dot g^i_l
=\int_\Om\{-\D wg^l_i+w^l_i\}\dot g^i_l
\end{aligned}
\end{equation}
and conclude
\begin{equation}
\dde{(Rw\sqrt g)}{g_{ij}}=(\tfrac12 Rg^{ij}-R^{ij})w\sqrt g+(w^{ij}-\D wg^{ij})\sqrt g.
\end{equation}
\ep

\section{The quantization} \las{4}
For the quantization of the Hamiltonian setting we use the same approach as in our former paper \cite{cg:qgravity}, at least in the beginning: First, we replace all densities by tensors, by choosing a fixed Riemannian metric in $\so$
\begin{equation}\lae{4.1} 
\chi=(\chi_{ij}(x)),
\end{equation}
and, for a given metric $g=(g_{ij}(t,x))$, we define
\begin{equation}
\f=\f(x,g_{ij})=\big(\frac{\det g_{ij}}{\det \chi_{ij}}\big)^\frac12
\end{equation}
such that the Einstein-Hilbert functional $J$ in \fre{3.17} can be written in the form
\begin{equation}
J=\int_a^b\int_\Om\{\frac14G^{ij,kl}\dot g_{ij}\dot g_{kl}w^{-2}+(R-2\Lam)\}w\f\sqrt \chi.
\end{equation}
The Hamilton density $\mc H$ is then replaced by the function
\begin{equation}
H=\{\f^{-1}G_{ij,kl}\pi^{ij}\pi^{kl}-(R-2\Lam)\f\}w,
\end{equation}
where now
\begin{equation}
\pi^{ij}=-\f G^{ij,kl}h_{kl}
\end{equation}
and
\begin{equation}
h_{ij}=-\f^{-1}G_{ij,kl}\pi^{kl}.
\end{equation}
The effective Hamiltonian is of course
\begin{equation}
w^{-1}H.
\end{equation}
Fortunately, we can, at least locally, assume
\begin{equation}
w=1
\end{equation}
by choosing an appropriate coordinate system: Let $(t_0,x_0)\in N$ be an arbitrary point, then consider the Cauchy hypersurface
\begin{equation}
M(t_0)=\{t_0\}\times \so
\end{equation}
and look at a tubular neighbourhood of $M(t_0)$, i.e., we define new coordinates $(t,x^i)$, where $(x^i)$ are coordinate for $\so$ near $x_0$ and $t$ is the signed Lorentzian distance to $M(t_0)$ such that the points
\begin{equation}
(0,x^i)\in M(t_0).
\end{equation}
The Lorentzian metric of the ambient space then has the form
\begin{equation}
d\bar s^2=-dt^2+g_{ij} dx^idx^j.
\end{equation}

Secondly, we use the same model as in \cite[Section 3]{cg:qgravity}: The Riemannian metrics $g_{ij}(t,\cdot)$ are elements of the bundle $T^{0,2}(\so)$. Denote by $E$ the fiber bundle with base $\so$ where the fibers consists of the Riemannian metrics $(g_{ij})$. We shall consider each fiber to be a Lorentzian manifold equipped with the DeWitt metric. Each fiber $F$ has dimension
\begin{equation}
\dim F=\frac{n(n+1)}2\equiv m+1.
\end{equation}
Let $(\xi^a)$, $0\le a\le m$, be  coordinates for a local trivialization such that
\begin{equation}
g_{ij}(x,\xi^a)
\end{equation}
is a local embedding. The DeWitt metric is then expressed as
\begin{equation}
G_{ab}=G^{ij,kl}g_{ij,a}g_{kl,b},
\end{equation}
where a comma indicates partial differentiation.  The Hamiltonian is then expressed as
\begin{equation}
H=\f^{-1}G^{ab}\pi_a\pi_b-(R-2\Lam)\f,
\end{equation}
\cf \cite[equ. (3.55)]{cg:qgravity}. The fibers equipped with the metric
\begin{equation}\lae{4.16}
(\f G_{ab})
\end{equation}
are then globally hyperbolic Lorentzian manifolds. The hypersurfaces
\begin{equation}
\{\f=\const\}
\end{equation}
are Cauchy hypersurfaces.

Let $F=F(x)$ be a fiber and set
\begin{equation}
\tau=\log\f,
\end{equation}
then $\tau$ is a time function. In the Gaussian coordinate system $(\tau,\xi^A)$, $1\le A\le m$, corresponding to the hypersurface
\begin{equation}\lae{4.19}
M=\{\f=1\}=\{\tau=0\}
\end{equation}
the metric \re{4.16} has the form
\begin{equation}\lae{4.20}
ds^2=\frac{4(n-1)}n\f\{-d\tau^2+G_{AB}d\xi^Ad\xi^B\}.
\end{equation}
where the Riemannian metric $G_{AB}$ is independent of $\tau$
\begin{equation}
\pde{G_{AB}}\tau=0.
\end{equation}
When we work in a local trivialization of $E$, the coordinates $\xi^A$ are independent of $x$.
\bl\lal{4.1}
The function $\f$ is independent of $x$.
\el
\bp
Let
\begin{equation}\lae{4.22}
g_{ij}(x,\tau,\xi^A)
\end{equation}
be the local embedding in $E$, then we have
\begin{equation}
\dot g_{ij}=\pde{g_{ij}}\tau=\frac2n g_{ij},
\end{equation}
\cf \cite[equ. (4.13)]{cg:qgravity}, hence we conclude
\begin{equation}\lae{4.24}
\begin{aligned}
g_{ij}&=e^{\frac2n\tau}g_{ij}(x,0,\xi^A)\\
&\equiv e^{\frac2n\tau}\s_{ij}(x,\xi^A),
\end{aligned}
\end{equation}
where
\begin{equation}
\s_{ij}=g_{ij}(0)\in M
\end{equation}
and we further deduce
\begin{equation}\lae{4.26}
\f^2=\frac{\det g_{ij}}{\det \chi_{ij}}=e^{2\tau}\frac{\det\s_{ij}}{\det\chi_{ij}}.
\end{equation}
In the embedding \re{4.22} $\tau$ is considered to be independent of $x$ being the time component of a coordinate system satisfying \re{4.19} and \re{4.20}. Therefore, we infer from \re{4.26} 
\begin{equation}\lae{4.27} 
\det \s_{ij}=\det \chi_{ij},
\end{equation}
proving the lemma.
\ep

We can now quantize the Hamiltonian setting using the original variables $g_{ij}$ and $\pi^{ij}$. We consider the bundle $E$ equipped with the metric \re{4.20}, or equivalently,
\begin{equation}\lae{4.28}
(\f G^{ij,kl}),
\end{equation}
which is the \tit{covariant} form, in the fibers and with the Riemannian metric $\chi$ in $\so$. Furthermore, let
\begin{equation}
C^\un_c(E)
\end{equation}
be the space of real valued smooth functions with compact support in $E$.

In the quantization process, where we choose $\hbar=1$, the variables $g_{ij}$ and $\pi^{ij}$ are then replaced by operators $\hat g_{ij}$ and $\hat\pi^{ij}$ acting in $C^\un_c(E)$ satisfying the commutation relations
\begin{equation}
[\hat g_{ij},\hat\pi^{kl}]=i\de^{kl}_{ij},
\end{equation}
while all the other commutators vanish. These operators are realized by defining $\hat g_{ij}$ to be the multiplication operator
\begin{equation}
\hat g_{ij} u=g_{ij}u
\end{equation}
and $\hat \pi^{ij}$ to be the \tit{functional} differentiation
\begin{equation}
\hat\pi^{ij}=\frac1i \dde{}{g_{ij}},
\end{equation}
i.e., if $u\in C^\un_c(E)$, then
\begin{equation}
\dde{u}{g_{ij}}
\end{equation}
is the Euler-Lagrange operator of the functional
\begin{equation}
\int_{\so}u\sqrt\chi\equiv\int_\so u.
\end{equation}
Hence, if $u$ only depends on $(x,g_{ij})$ and not on derivatives of the metric, then
\begin{equation}
\dde{u}{g_{ij}}=\pde u{g_{ij}}.
\end{equation}
Therefore, the transformed Hamiltonian $\hat H$ can be looked at as the hyperbolic differential operator
\begin{equation}\lae{4.36}
\hat H=-\D-(R-2\Lam)\f,
\end{equation}
where $\D$ is the Laplacian of the metric in \re{4.28} acting on functions
\begin{equation}
u=u(x,g_{ij}).
\end{equation}
We used this approach in \cite{cg:qgravity} to transform the Hamilton constraint to the Wheeler-DeWitt equation
\begin{equation}
\hat Hu=0\qq \text{in } E
\end{equation}
which can be solved with suitable Cauchy conditions. However, the Hamiltonian in the Wheeler-DeWitt equation is a differential operator that only acts in the fibers of $E$ and not in the base space $\so$ which seems to be insufficient. This short-coming will be eliminated when, instead of the explicit Hamilton constraint, its equivalent  implicit version, equation \fre{3.33} is quantized: Following Dirac the Poisson brackets are replaced by $\frac1i$ times the commutators in the quantization process, $\hbar=1$, i.e., we obtain
\begin{equation}
\{\pi^{ij},H\}\ra i[\hat H, \hat\pi^{ij}].
\end{equation}
Dropping the hats in the following to improve the readability equation \re{3.33} is transformed to
\begin{equation}\lae{4.40}
\begin{aligned}
ig_{ij}[H,\pi^{ij}]&=(n-1) (R-2\Lam)\f-R  \f+\D,
\end{aligned}
\end{equation}
where $\D$ is the Laplace operator with respect to the fiber metric.

Now, we have
\begin{equation}\lae{4.41}
\begin{aligned}
i[H,\pi^{ij}]&=[H,\dde{}{g_{ij}}]\\
&= [-\D,\dde{}{g_{ij}}]-[(R-2\Lam)\f,\dde{}{g_{ij}}],
\end{aligned}
\end{equation}
\cf \re{4.36}. Since we apply both sides to functions $u\in C^\un_c(E)$ 
\begin{equation}\lae{4.42}
[-\D,\dde{}{g_{ij}}]u=[-\D,\pde{}{g_{ij}}]u=-R^{ij}_{\hp{ij},kl}u^{kl},
\end{equation}
because of the Ricci identities, where
\begin{equation}
R^{ij}_{\hp{ij},kl}
\end{equation}
is the Ricci tensor of the fiber metric \re{4.28} and
\begin{equation}
u^{kl}=\pde u{g_{kl}}
\end{equation}
is the gradient of $u$.

For the second commutator on the right-hand side of \re{4.41} we obtain
\begin{equation}\lae{4.45}
\begin{aligned}
-[(R-2\Lam)\f,\dde{}{g_{ij}}]u=-(R-2\Lam)\f\pde u{g_{ij}}+\dde{}{g_{ij}}\{(R-2\Lam)u\f\},
\end{aligned}
\end{equation}
where the last term is the Euler-Lagrange operator of the functional
\begin{equation}
\begin{aligned}
\int_\so (R-2\Lam)u\f&\equiv \int_\so(R-2\Lam)u\f\sqrt\chi\\
&=\int_\so(R-2\Lam)u\sqrt g
\end{aligned}
\end{equation}
with respect to the variable $g_{ij}$, since the scalar curvature $R$ depends on the derivatives of $g_{ij}$. From \re{4.11.1} and the proof of \frl{3.3} we infer  
\begin{equation}\lae{4.47} 
\begin{aligned}
\dde{}{g_{ij}}\{(R-2\Lam)u\f\}&=\frac12 (R-2\Lam) g^{ij}u\f-R^{ij}u\f\\
&+\f\{u_{;}^{\hp;ij}-\tilde\D u g^{ij}\}+(R-2\Lam)\f\pde u{g_{ij}},
\end{aligned}
\end{equation}
where the semicolon indicates covariant differentiation in $\so$ with respect to the metric $g_{ij}$, $\tilde \D$ is the corresponding Laplacian, and where we observe that the fundamental lemma of the calculus of variations has been applied to functions in $L^2(\so,\sqrt\chi)$, i.e.,
\begin{equation}
\int_\so f\h\sqrt g=\int_\so f\h\f\sqrt\chi;
\end{equation}
here we have 
\begin{equation}
f\in C^0(\so),\q \h\in C^\un_c(\so).
\end{equation}
We also note that
\begin{equation}
\begin{aligned}
D_ku&= \pde u{x^k}+\pde u{g_{ij}}\pde{g_{ij}}{x^k}\\
&=\pde u{x^k}
\end{aligned}
\end{equation}
in Riemannian normal coordinates.

Hence, we conclude that equation \re{4.40} is equivalent to
\begin{equation}\lae{4.59}
\begin{aligned}
-\D u -(n-1)\f\tilde \D u-\frac {n-2}2\f(R-2\Lam)u=0
\end{aligned}
\end{equation}
in $E$, since
\begin{equation}
g_{ij}R^{ij}_{\hp{ij},kl}=0
\end{equation}
for
\begin{equation}\lae{4.61}
\frac1{\sqrt{n(n-1)\f}}g_{ij}
\end{equation}
is the future directed unit normal of the Cauchy hypersurfaces $\{\f=\const\}$: The gradient of $\f$
\begin{equation}
\pde\f{g_{ij}}=\frac12\f g^{ij}
\end{equation}
is a past directed normal in \tit{covariant} notation. Its contravariant version has the form
\begin{equation}
\f^{-1}G_{ij,kl}g^{kl}\frac12\f=-\frac1{2(n-1)}g_{ij}.
\end{equation}
Therefore, the vector in \re{4.61} is future directed and has unit length as can easily be checked.

Now, let us choose a coordinate system $(\tau,\xi^A)$ associated with the Cauchy hypersurface
\begin{equation}
M=\{\f=1\}
\end{equation}
and express the metric as in \re{4.20}. The time coordinate $\tau$ is defined as
\begin{equation}
\tau=\log\f.
\end{equation}
Let $t$ be the time function
\begin{equation}\lae{4.58}
t=\sqrt\f=e^{\frac12\tau},
\end{equation}
then
\begin{equation}
dt^2=\frac14\f d\tau^2
\end{equation}
and we conclude that the fiber metric can be expressed as
\begin{equation}\lae{4.68}
ds^2=-\frac{16(n-1)}ndt^2+\frac{4(n-1)}nt^2 G_{AB}d\xi^Ad\xi^B,
\end{equation}
where $G_{AB}$ is independent of $t$. We also emphasize that $t$ is independent of $x$, \cf \rl{4.1}. 

Let $(\xi^a)=(t,\xi^A)$, $0\le a\le m$, be the coordinates such that
\begin{equation}
\xi^0=t\q\wed\q 1\le A\le m,
\end{equation}
then we immediately deduce from \re{4.68} or \re{4.20} that the Ricci tensor satisfies
\begin{equation}
R_{0a}=0\qq\A\, 0\le a\le m.
\end{equation}
Since the determinant of the metric in \re{4.68} is equal to
\begin{equation}
\abs{\det(G_{ab})}=16(\tfrac {n-1}n)\{4(\tfrac {n-1}n)\}^mt^{2m}\det(G_{AB})
\end{equation}
 we conclude that the equation \re{4.59} can be expressed in the form
\begin{equation}\lae{4.74}
\begin{aligned}
&\frac1{16}\frac{n}{n-1}t^{-m}\pde{(t^m\dot u)}t -\frac14 \frac n{n-1}t^{-2}\D_Gu\\
&\qq\qq-(n-1) t^2 \tilde\D u-\frac {n-2}2t^2(R-2\Lam)u=0,
\end{aligned}
\end{equation}
where $\D_G$ is the Laplacian with respect to the metric $G_{AB}$.

For any point
\begin{equation}
(x,g_{ij})\in E
\end{equation}
the metric can be written in the form
\begin{equation}
g_{ij}=t^\frac4n\s_{ij},
\end{equation}
where $\s_{ij}$ is independent of $t$ and 
\begin{equation}
\det \s_{ij}=\det \chi_{ij},
\end{equation}
\cf \re{4.24} and \re{4.27}. Hence, we can write
\begin{equation}
\tilde \D u =t^{-\frac4n}\tilde\D_{\s_{ij}}u.
\end{equation}
Thus, equipping $E$ with the metric 
\begin{equation}\lae{4.79}
\begin{aligned}
d\bar s^2&=-\frac{16(n-1)}{n}dt^2+\frac{4(n-1)}nt^2 G_{AB}d\xi^Ad\xi^B+\frac1{n-1}\s_{ij}dx^idx^j\\
&\equiv G_{ab}d\xi^ad\xi^b+\frac1{n-1}\s_{ij}dx^idx^j\\
&\equiv G_{\al\bet}d\zeta ^\al d\zeta^\bet,
\end{aligned}
\end{equation}
where $0\le a\le m$ and $\xi^0=t$. We call $G_{ab}$ the fiber metric and $\s_{ij}$ the base metric, which are to be evaluated at the points 
\begin{equation}
(x,\xi^a)\equiv (x,g_{ij})=(x,t^\frac4n\s_{ij}).
\end{equation}
Beware that
\begin{equation}
\s_{ij}=\s_{ij}(x,\xi^A)\in E_1,
\end{equation}
where $E_1$ is the subbundle
\begin{equation}
E_1=\{t=1\}.
\end{equation}
 This metric the operator $P$ in \re{4.74} is a symmetric hyperbolic differential operator
\begin{equation}
Pu=-D_\al(a^{\al\bet}D_\bet u),
\end{equation}
where the derivatives are covariant derivatives with respect to the metric in \re{4.79} and the coefficients $a^{\al\bet}$ represent a Lorentzian metric. However, it is not  normally hyperbolic, i.e., its main part is not identical with the Laplacian of the ambient metric. Nevertheless, we can consider $P$ as a normally hyperbolic operator by equipping $E$ with the metric
\begin{equation}
\begin{aligned}
d\tilde s^2&=-\frac{16(n-1)}{n}dt^2+\frac{4(n-1)}nt^2 G_{AB}d\xi^Ad\xi^B\\
&\msp[140]+\frac1{n-1}t^{\frac4n-2}\s_{ij}dx^idx^j\\
&\equiv \tilde G_{\al\bet}d\zeta^\al d\zeta^\bet,
\end{aligned}
\end{equation}
though, of course, $P$ is not symmetric in this metric. 

Let $E$, $\tilde E$ be the bundles
\begin{equation}
(E,G_{\al\bet})\q\wed\q (E,\tilde G_{\al\bet})
\end{equation}
respectively, and $E_1$ \resp $\tilde E_1$ the corresponding subbundles defined by
\begin{equation}
\{t=1\}.
\end{equation}
We shall now prove that $E$ and $\tilde E$ are both globally hyperbolic manifolds and the subbundles $E_1$ \resp $\tilde E_1$, or more generally, the subbundles $E_1(\tau)$ \resp $\tilde E_1(\tau)$, defined by
\begin{equation}
\{t=\tau\},\q\tau>0,
\end{equation}
Cauchy hypersurfaces provided the base space $\so$ is either compact or a homogeneous space for a suitable metric $\rho_{ij}$.

\bl\lal{4.2}
The bundles $E$ and $\tilde E$ are both globally hyperbolic manifolds, if $\so$ is either compact or a homogeneous space for a suitable metric $\rho_{ij}$, and the hypersurfaces $E_1(\tau)$ \resp $\tilde E_1(\tau)$ are Cauchy hypersurfaces.
\el
\bp
We shall only prove that $E$ is globally hyperbolic, since the proof for $\tilde E$ is essentially identical. We shall show that $E_1$ is a Cauchy hypersurface. The arguments will then also apply in case of the hypersurfaces $E_1(\tau)$. The proof will be similar to the proof of \cite[Lemma 4.3]{cg:qgravity}, where we proved that the fibers of $E$ are globally hyperbolic. The fact that we now consider  the whole bundle creates a small complication which will be handled by the additional assumption on $\so$.

We shall now prove that $E_1$ is a Cauchy hypersurface implying that $E$ is globally hyperbolic. Let us argue by contradiction. Thus, let 
\begin{equation}
\ga(s)=(\ga^\al(s)),\q s\in I=(a,b),
\end{equation}
be an inextendible future directed causal curve in $E$ and assume that $\ga$ does not intersect $E_1$. We shall show that this will lead to a contradiction. It is also obvious that $\ga$ can meet $E_1$ at most once.

Assume that there exists $s_0\in I$ such that
\begin{equation}
t(\ga(s_0))<1
\end{equation}
and assume from now on that $s_0$ is the left end point of $I$. Since $t$ is continuous, the whole curve $\ga$ must be contained in the past of $E_1$.

$\ga$ is causal, i.e.,
\begin{equation}
\begin{aligned}
\frac1{n-1}\s_{ij}\dot x^i\dot x_j+\frac{4(n-1)}nt^2G_{AB}\dot\ga^A\dot\ga^B\le \frac{16(n-1)}{n}\abs{\dot\ga^0}^2
\end{aligned}
\end{equation}
and thus
\begin{equation}
\sqrt{\frac1{n-1}\s_{ij}\dot x^i\dot x_j+\frac{4(n-1)}nt^2G_{AB}\dot\ga^A\dot\ga^B}\le 4\dot\ga^0,
\end{equation}
since $\ga$ is future directed.

Let
\begin{equation}
\tilde\ga=(x^i,\ga^A)
\end{equation}
be the projection of $\ga$ onto $E_1$, then the length of $\tilde\ga$ is bounded
\begin{equation}
\begin{aligned}
L(\tilde\ga)&\le \int_I\sqrt{\frac1{n-1}\s_{ij}\dot x^i\dot x_j+\frac{4(n-1)}nG_{AB}\dot\ga^A\dot\ga^B}\\
&\le 4(1-t(s_0))<4.
\end{aligned}
\end{equation}
Expressing the quadratic form
\begin{equation}
G_{AB}\dot\ga^A\dot\ga^B
\end{equation}
in $E_1$ in the coordinates $(g_{ij})=(\s_{ij})$, we have
\begin{equation}
\begin{aligned}
G_{AB}\dot\ga^A\dot\ga^B&=\s^{ik}\s^{jl}\dot\s_{ij}\dot\s_{kl}\\
&\equiv\norm{\dot\s_{ij}}^2,
\end{aligned}
\end{equation}
since the right-hand side is exactly
\begin{equation}
G^{ij,kl}\dot\s_{ij}\dot\s_{kl},
\end{equation}
if
\begin{equation}
\dot\s_{ij}\in T(E_1).
\end{equation}
Hence, we infer, in view of \cite[Lemma 14.2]{hamilton:ricci}, that the metrics $(\s_{ij}(s))$ are all uniformly equivalent in $I$ and converge to a positive definite metric when $s$ tends to $b$. It remains to prove that the points $(x^i(s))$ are precompact in $\so$, then we would have derived a contradiction.

If $\so$ is compact then the precompactness of $(x^i(s))$ is trivial, thus let us assume that $(\so, \rho_{ij})$ is a homogeneous space. Then $\s_{ij}(s_0)$ is equivalent to $\rho_{ij}(x(s_0))$ and hence, in view of the homogeneity, $\s_{ij}(s)$ is uniformly equivalent to $\rho_{ij}(x(s))$ for all $s\in I$, and we conclude
\begin{equation}
\int_I\sqrt{\rho_{ij}\dot x^i\dot x^j}\le \const
\end{equation}
proving the precompactness. $E_1$ is therefore a Cauchy hypersurface and $E$ is globally hyperbolic.
\ep

\br
Since $\tilde E$ is globally hyperbolic and $P$ is a normally hyperbolic differential operator the Cauchy problems
\begin{equation}\lae{4.100}
\begin{aligned}
Pu&=f,\\
\fv u{\tilde E_1(\tau)}&=u_0,\\
\fv {u_\al\tilde\nu^\al}{\tilde E_1(\tau)}&=u_1
\end{aligned}
\end{equation}
have unique solutions
\begin{equation}
u\in C^\un(\tilde E)
\end{equation}
for given values $u_0,u_1\in C^\un_c(\tilde E_1(\tau))$ and $f\in C^\un_c(\tilde E)$ such that
\begin{equation}
\supp u\su J^{\tilde E}(K),
\end{equation}
where
\begin{equation}
K=\supp u_0\uu \supp u_1\uu\supp f,
\end{equation}
\cf \cite{gunther:book, baer:book, ginoux:fredenhagen}.
\er

Since $E$, $\tilde E$ and $E_1(\tau)$ \resp $\tilde E_1(\tau)$ coincide as sets and the normals $(\nu^\al)$ \resp $\tilde\nu^\al)$ are also identical
\begin{equation}
\tilde\nu=\nu
\end{equation}
we immediately deduce that the Cauchy problems \re{4.100} are also uniquely solvable in $E$. Using this information we then could derive the existence of the fundamental solutions $F_\pm$ for $P$ in $E$ and also the existence of the advanced \resp retarded Green's operators $G_\pm$ of $P$, \cf \cite[Theorem 4]{ginoux:fredenhagen}.

However, we would like to show how the fundamental solutions $\tilde F_\pm$ of $P$ in $\tilde E$ can easily be transformed to yield fundamental solutions of $P$ in $E$ and similarly  the Green's functions $\tilde G_\pm$. This process is valid in general pseudo-riemannian manifolds, and thus also valid for elliptic operators, however, we shall only consider Lorentzian manifolds. The notations $N$ \resp $\tilde N$ refer to the same manifold $N$ equipped with the metrics $g_{\al\bet}$ \resp $\tilde g_{\al\bet}$.

\bd
Let $T\in \mc D'(N)$ be a distribution and let $\sqrt {\abs g}$ be the volume element in $N$, where
\begin{equation}
g=\det g_{\al\bet},
\end{equation}
then we use the notation
\begin{equation}
\spd T{\h\sqrt{\abs g}}
\end{equation}
or
\begin{equation}
T[\h\sqrt{\abs g}]
\end{equation}
to refer to \cq{T acts on $\h$} instead of the usual symbols
\begin{equation}
\spd T\h
\end{equation}
or
\begin{equation}
T[\h].
\end{equation}
If $P$ is a differential operator in $N$ and $P^*$ its formal adjoint, then
\begin{equation}
\spd{PT}{\h\sqrt{\abs g}}=\spd T{(P^*\h)\sqrt{\abs g}}.
\end{equation}
\ed
We found this notation in \cite[Definition 2.8.1, p. 60]{friedlander:book}

\bl
Let $T\in \mc D'(N,\tilde g)$ and let $g$ be a another smooth metric in $N$ and set
\begin{equation}
\psi=\frac{\sqrt{\abs{\tilde g}}}{\sqrt{\abs g}},
\end{equation}
then
\begin{equation}
\psi T\in \mc D'(N,g)
\end{equation}
and
\begin{equation}
\spd{\psi T}{\h\sqrt{\abs g}}=\spd T{\h\sqrt{\abs{\tilde g}}}\qq\A\,\h\in C^\un_c(N).
\end{equation}
\el
\bp
Follows immediately from the definition of $\psi T$
\begin{equation}
\spd{\psi T}{\h\sqrt{\abs g}}=\spd T{\psi\h\sqrt{\abs g}}=\spd T{\h\sqrt{\abs{\tilde g}}}.
\end{equation}
\ep

As an application we obtain:
\bc\lac{4.6}
Let $\tilde F_\pm$ \resp $\tilde G_\pm$ be the fundamental solutions of $P$ in $\tilde E$ \resp the advanced and retarded Green's operators, and define
\begin{equation}
\psi=\frac{\sqrt{\abs{\tilde G}}}{\sqrt{\abs G}}=t^{2-n},
\end{equation}
then
\begin{equation}\lae{4.116}
F_\pm=\psi\tilde F_\pm
\end{equation}
are fundamental solutions of $P$ in $E$ and
\begin{equation}\lae{4.117}
G_\pm=\psi\tilde G_\pm
\end{equation}
the advanced and retarded Green's operators.
\ec
\bp
\cq{\re{4.116}}\q We have
\begin{equation}\lae{4.118}
\begin{aligned}
F_\pm[\h\sqrt{\abs G}\,]&=\psi\tilde F_\pm[\h\sqrt{\abs G}\,]\\
&=\tilde F_\pm[\h\sqrt{\abs{\tilde G}}\,]
\end{aligned}
\end{equation}
and
\begin{equation}
PF_\pm[\h\sqrt{\abs G}\,]=P\tilde F_\pm[\h\sqrt{\abs{\tilde G}}\,]=\h.
\end{equation}

\cvm
\cq{\re{4.117}}\q To prove the second claim we note that the Green's operators are defined as maps
\begin{equation}
C^\un_c(E)\ra C^\un(E)
\end{equation}
by the definition
\begin{equation}
G_\pm[\h\sqrt{\abs G}\,](p)=F_\pm(p)[\h\sqrt{\abs G}\,],\qq p\in E.
\end{equation}
Now, from \re{4.118} we deduce
\begin{equation}
\begin{aligned}
F_\pm(p)[\h\sqrt{\abs G}\,]&=\tilde F_\pm(p)[\h\sqrt{\abs{\tilde G}}\,]\\
&=\tilde G_\pm[\h\sqrt{\abs{\tilde G}}\,](p)\\
&=\psi\tilde G_\pm[\h\sqrt{\abs G}\,](p).
\end{aligned}
\end{equation}
\ep

\br\lar{4.7}
Let $G$ be the Green's operator of $P$ in $E$
\begin{equation}
G=G_+-G_-,
\end{equation}
then
\begin{equation}
N(P)=\{Gu:u\in C^\un_c(E)\}
\end{equation}
is the kernel of $P$. Its elements are smooth functions which are spacelike compact; however, this condition is  strictly correct only in $\tilde E$, since the light cones in $\tilde E$ and $E$ are different. Fortunately, we only need one special property of spacelike compact functions, namely, that their restrictions to Cauchy hypersurfaces have compact support. This will be case in $E$, if we only consider the Cauchy hypersurfaces $E_1(\tau)$, as we shall prove in the lemma below.
\er

\bl
The compact subsets of $\tilde E_1(\tau)$ are also compact in $E_1(\tau)$ and vice versa.
\el
\bp
The Cauchy hypersurfaces $E_1(\tau)$ \resp $\tilde E_1(\tau)$ carry the same topology, since their induced metrics are uniformly equivalent as one easily checks.
\ep

\section{The second quantization}\las{5}
Let us first summarize some facts about the Green's operators $G_\pm$ of $P$ in $E$ which are still valid even though $P$ is not normally hyperbolic.
\bl
Let $G_\pm$ \resp $\tilde G_\pm$ be the Green's operators of $P$ in $E$ \resp $\tilde E$, then
\begin{equation}\lae{5.1}
G_\pm:C^\un_c(E)\ra C^\un(E)
\end{equation}
\begin{equation}\lae{5.2}
P\circ G_\pm=\fv{G_\pm\circ P}{C^\un_c(E)}=\fv{\id}{C^\un_c(E)}
\end{equation}
\begin{equation}\lae{5.3}
\supp(G_\pm u)=\supp(\tilde G_\pm u)\q\A\,u\in C^\un_c(E)
\end{equation}
\begin{equation}\lae{5.4}
\supp G_+u\su J_+^{\tilde  E}(\supp u)\q \A\,u\in C^\un_c(E)
\end{equation}
\begin{equation}\lae{5.5}
\supp G_-u\su J_-^{\tilde  E}(\supp u)\q \A\,u\in C^\un_c(E)
\end{equation}
\begin{equation}\lae{5.6}
\supp G_+u\ii\supp G_-v \q \text{is compact}
\end{equation}
for all $u,v\in C^\un_c(E)$, and
\begin{equation}\lae{5.7} 
G_\pm^*=G_\mp.
\end{equation}
\el
\bp
The properties \re{5.1} and \re{5.2} immediately follow from the corresponding relations for $\tilde G\pm$ of $P$ in $\tilde E$ and the fact that
\begin{equation}
G_\pm=t^{2-n}\tilde G_\pm,
\end{equation}
\cf \frc{4.6}. The preceding relation also proves the properties \re{5.3}, \re{5.4}, \re{5.5} and \re{5.6}, since the topologies of $E$ and $\tilde E$ are identical. 

It remains to prove \re{5.7}. Let $u,v\in C^\un_c(E)$, then
\begin{equation}
\begin{aligned}
\int_E\spd{G_\pm u}v&=\int_E\spd{G_\pm u}{PG_\mp v}\\ 
&= \int_E\spd{PG_\pm u}{G_\mp v}\\
&=\int_E \spd{u}{G_\mp v},
\end{aligned}
\end{equation}
where the partial integration is justified because of \re{5.6}, and the scalar product is just normal multiplication.
\ep
\bl
Let $E_1(\tau)$ be one of the special Cauchy hypersurfaces in $E$, then
\begin{equation}\lae{5.10}
\begin{aligned}
\int_E\spd{u}{Gv}=\int_{E_1(\tau)}\{\spd{D_\nu(Gu)}{Gv}-\spd{Gu}{D_\nu Gv}\},
\end{aligned}
\end{equation}
for all $u,v\in C^\un_c(E)$, where $\nu$ is the future directed normal of $E_1(\tau)$.
\el
\bp
Let $E_+$, $E_-$ be defined by
\begin{equation}
E_+=\{t>\tau\}
\end{equation}
and
\begin{equation}
E_-=\{t<\tau\},
\end{equation}
then
\begin{equation}
\int_E\spd u{Gv}=\int_{E_+}\spd u{Gv}+\int_{E_-}\spd u{Gv}.
\end{equation}
Now, in $E_+$ we have
\begin{equation}
PG_-u=u
\end{equation}
and
\begin{equation}
PGv=0=GPv.
\end{equation}
Moreover, 
\begin{equation}
\supp(G_- u)\ii E_+\q\text{is compact},
\end{equation}
since
\begin{equation}
\supp(\tilde G_- u)\ii \tilde E_+\q\text{is compact},
\end{equation}
hence we obtain by partial integration
\begin{equation}
\begin{aligned}
\int_{E_+}\spd{PG_-u}{Gv}=-\int_{E_1(\tau)}\spd{D_\nu G_-u}{Gv}+\int_{E_1(\tau)}\spd{G_-u}{D_\nu Gv}.
\end{aligned}
\end{equation}
A similar argument applies to  $E_-$ by looking at
\begin{equation}
PG_+u=0
\end{equation}
leading to 
\begin{equation}
\begin{aligned}
\int_{E_-}\spd{PG_+u}{Gv}=\int_{E_1(\tau)}\spd{D_\nu G_+u}{Gv}-\int_{E_1(\tau)}\spd{G_+u}{D_\nu Gv}.
\end{aligned}
\end{equation}
Adding these two equations implies the result.
\ep

We shall now construct a CCR representation or a Weyl system for $P$ and its kernel
\begin{equation}
N(P)=\{u\in C^\un(E):Pu=0\}=\{Gu:u\in C^\un_c(E)\}.
\end{equation}
This characterization of $N(P)$ is correct, since it is valid in $\tilde E$ and because of
\begin{equation}
PG[u\sqrt{\abs G}\,]=P\tilde G[u\sqrt{\abs{\tilde G}}\,],
\end{equation}
\cf \fre{4.117}.


There are two ways to construct a Weyl system given a formally self-adjoint, normally hyperbolic operator in a globally hyperbolic spacetime which are also applicable in our case, though $P$ is not normally hyperbolic. One possibility is to define a symplectic vector space
\begin{equation}\lae{5.23}
V=C^\un_c(e)/N(G),
\end{equation}
where $G$ is the Green's operator of $P$
\begin{equation}
G=G_+-G_-.
\end{equation}
Since
\begin{equation}\lae{5.25}
G^*=-G
\end{equation}
the bilinear form
\begin{equation}\lae{5.26}
\om(u,v)=\int_E\spd{u}{Gv},\qq u,v\in V,
\end{equation}
is skew-symmetric, non-degenerate by definition, and hence symplectic. Then, there is a canonical way to construct a corresponding Weyl system.

The second method is to pick a Cauchy hypersurface $E_1$ in $E$ and then define a quantum field $\F$ with values in the space of essentially self-adjoint operators in a corresponding symmetric Fock space.

We pick a Cauchy hypersurface $E_1=E_1(\tau)$ in $E$ and define the complex Hilbert space
\begin{equation}
H_{E_1}=L^2(E_1)\otimes \Cc=L^2(E_1,\Cc)
\end{equation}
the complexification of the real Hilbert space $L^2(E_1)$ with complexified scalar product
\begin{equation}
\spd uv_{E_1}=\int_{E_1}\spd uv_{\Cc}.
\end{equation}
We denote the symmetric Fock space of $H_{E_1}$ by $\mc F(H_{E_1})$. Let $\Theta$ be the corresponding Segal field. Since $G^*=-G$, we deduce from \re{5.4}, \re{5.6} and \frr{4.7} that
\begin{equation}
\fv{G^*u}{E_1}\in C^\un_c(E_1)\su H_{E_1}\q\A\, u\in C^\un_c(E).
\end{equation}
We can therefore define
\begin{equation}\lae{5.30}
\F_{E_1}(u)=\Theta(i\fv{(G^*u)}{E_1}-\fv{D_\nu(G^*u)}{E_1}).
\end{equation}
From the proof of \cite[Lemma 4.6.8]{baer:book} we conclude that the right-hand side of \re{5.30} is an essentially self-adjoint operator in $\mc F(H_{E_1})$.  We therefore call the map $\F_{E_1}$ from $C^\un_c(E)$ to the set of self-adjoint operators in $\mc F(H_{E_1})$ a quantum field defined in $E_1$.
\bl
The quantum field $\F_{E_1}$ satisfies the equation
\begin{equation}
P\F_{E_1}=0
\end{equation}
in the distributional sense, i.e.,
\begin{equation}
\begin{aligned}
\spd{P\F_{E_1}}u&=\spd{\F_{E_1}}{Pu}\\
&=\F_{E_1}(Pu)=0\qq\A\,u\in C^\un_c(E).
\end{aligned}
\end{equation}
\el
\bp
In view of \re{5.25} we have
\begin{equation}
G^*(Pu)=0.
\end{equation}
\ep
With the help of the quantum field $\F_{E_1}$ we shall construct a Weyl system and hence a CCR representation of the symplectic vector space $(V,\om)$ which we defined in \re{5.23} and \re{5.26}.

From \re{5.30} we conclude  the commutator relation
\begin{equation}
[\F_{E_1}(u),\F_{E_1}(v)]=i\imc\spd{iG^*u-D_\nu(G^*u)}{iG^*v-D_\nu(G^*v)}_{E_1}I,
\end{equation}
for all $u,v\in C^\un_c(E)$, \cf \cite[Proposition 5.2.3]{robinson:book2}, where both sides are defined in the algebraic Fock space $\mc F_{\tup{alg}}(H_{E_1})$. 

On the other hand   
\begin{equation}\lae{5.35}
\begin{aligned}
\imc\spd{iG^*u-D_\nu(G^*u)}{iG^*v-D_\nu(G^*v)}_{E_1}&\\
&\msp[-350]=-\imc\spd{iG^*u}{D_\nu(G^*v)}_{E_1}-\imc\spd{D_\nu(G^*u)}{iG^*v}_{E_1}\\
&\msp[-350]=\int_{E_1}\{\spd{G^*u}{D_\nu(G^*v)}-\spd{D_\nu(G^*u)}{G^*v}\}\\
&\msp[-350]=\int_{E}\spd u{Gv}
\end{aligned}
\end{equation}
in view of \re{5.10} and \re{5.25}.

As a corollary we conclude
\begin{equation}\lae{5.36}
[\F_{E_1}(u),\F_{E_1}(v)]=i\int_{E_1}\spd u{Gv}I\q\A\, u,v\in C^\un_c(E).
\end{equation}

From \cite[Proposition 5.2.3]{robinson:book2} and \re{5.35} we immediately infer
\bt\lat{5.6}
Let $(V,\om)$ be the symplectic vector space in \re{5.23} and \re{5.26} and denote by $[u]$ the equivalence classes in $V$, then
\begin{equation}
W([u])=e^{i\F_{E_1}(u)}
\end{equation}
defines a Weyl system for $(V,\om)$, where $\F_{E_1}(u)$ is now supposed to be the closure of $\F_{E_1}(u)$ in $\mc F(H_{E_1})$, i.e., $\F_{E_1}(u)$ is a self-adjoint operator. The Weyl system generates a $C^*$-algebra with unit which we call a CCR representation of $(V,\om)$.
\et
\br\lar{5.7}
Since all CCR representations of $(V,\om)$ are $^*$-isomorphic, where the isomorphism maps Weyl systems to Weyl systems, \cf \cite[Theorem 5.2.8]{robinson:book2},  this especially applies  to the CCR representations corresponding to different Cauchy hypersurfaces ${E_1=E_1(\tau)}$ and ${E_1}'=E_1(\tau')$, i.e., there exists a $^*$-isomorphism $T$ such that
\begin{equation}
T(e^{i\F_{E_1}(u)})=e^{i\F_{{E_1}'}(u)}\qq \A\, [u]\in V.
\end{equation}
\er
\section{The gravitational waves  model}\las{6}
In the previous sections we saw that the quantization of the Hamilton constraint does not yield a unique result but depends on the equation by which the Hamilton constraint is expressed. In \cite{cg:qgravity} we obtain the Wheeler-DeWitt equation  after quantization and in the previous sections the equation \fre{4.74} which differs significantly. In this section we shall propose yet another model by replacing any occurrence of the term
\begin{equation}\lae{6.1.1}
\abs A^2-H^2
\end{equation}
by
\begin{equation}
(R-2\Lam).
\end{equation}
However, when we do this on the right-hand side of \fre{3.33}, then after quantization, we would obtain an elliptic equation instead of an hyperbolic equation, namely,
\begin{equation}\lae{6.3.1} 
-(n-1)\tilde{\D} u +\frac{n-4}2(R-2\Lam)u=0 
\end{equation}
valid in $E$, which, for fixed $(t,g_{ij})$, can be looked at as an eigenvalue equation, where $\Lam$ would be a constant multiple of the eigenvalue provided $n\not=4$. In case $\so$ is compact, a spectral resolution of equation \re{6.3.1} would be possible.

However, we believe that a hyperbolic and not an elliptic equation should define the possible states of quantum gravity. In order to obtain a hyperbolic equation while eliminating any occurrences of the term in \re{6.1.1} we have to express the Hamilton constraint by a different equation. In \rs{3} the Hamilton equations only yielded the tangential Einstein equations \fre{3.22}, or equivalently,
\begin{equation}\lae{6.4.1}
\bar R_{ij}-\frac12\bar Rg_{ij}+\Lam g_{ij}=0.
\end{equation}
The Hamilton constraint expresses the normal component of the Einstein equations, where the terms \tit{tangential} und \tit{normal} refer to the foliation $M(t)$ of the spacetime $N$. This foliation is also the solution set of the geometric flow equation
\begin{equation}
\dot x=-w\nu
\end{equation}
with initial hypersurface 
\begin{equation}
M_0=\so,
\end{equation}
where $\nu$ is the past directed normal $\nu$ of the solution hypersurfaces $M(t)$, \cf \cite[equ. (2.3.25)]{cg:cp}. We shall use the evolution equation of the mean curvature $H(t)$ of the $M(t)$ to define the Hamilton constraint.  

The mean curvature satisfies the evolution equation
\begin{equation}\lae{6.7.1} 
\dot H= -\tilde\D w+\{\abs A^2+\bar R_{\al\bet}\nu^\al\nu^\bet\}w,
\end{equation}
where we embellished the Laplacian with a tilde, \cf \cite[equ. (2.3.27)]{cg:cp} observing that in that reference
\begin{equation}
e^\psi =w.
\end{equation}
To exploit this evolution equation we need the following lemma: 
\bl
Assume that the equation \re{6.4.1} is valid, then 
\begin{equation}\lae{3.30.6}
\tfrac12\bar R=\tfrac1{n-1}\{G_{\al\bet}\nu^\al\nu^\bet-\Lam\}+\tfrac{n+1}{n-1}\Lam
\end{equation}
and 
\begin{equation}\lae{3.31.6}
\bar R_{\al\bet}\nu^\al\nu^\bet=\tfrac{n-2}{n-1}\{G_{\al\bet}\nu^\al\nu^\bet-\Lam\}-\tfrac2{n-1}\Lam.
\end{equation}
\el
\bp
\cq{\re{3.30.6}}\q There holds
\begin{equation}
\bar R=g^{ij}\bar R_{ij}-\bar R_{\al\bet}\nu^\al\nu^\bet
\end{equation}
and hence
\begin{equation}\lae{3.33.6}
\bar R_{\al\bet}\nu^\al\nu^\bet +\tfrac 12\bar R=\tfrac{n-1}2\bar R-n\Lam
\end{equation}
or, equivalently,
\begin{equation}\lae{3.34.6}
\tfrac1{n-1}\{\ein-\Lam\}=\tfrac12\bar R-\tfrac{n+1}{n-1}\Lam.
\end{equation}
\cq{\re{3.31.6}}\q Combining  \re{3.33.6} and \re{3.34.6} we deduce
\begin{equation}
\rnn=\tfrac{n-2}{n-1}\{\ein-\Lam\}-\tfrac2{n-1}\Lam.
\end{equation}
\ep
We note that
\begin{equation}\lae{6.15.1} 
\pi^{ij}=(Hg^{ij}-h^{ij})\f,
\end{equation}
where $(h^{ij})$ is the contravariant version of the second fundamental form and where we also point out that, as before, we introduced the function $\f$ to replace the density $\sqrt g$ in order to deal with tensors instead of densities.

Hence, we have
\begin{equation}\lae{6.16.1}
(n-1) H\f=g_{ij}\pi^{ij}
\end{equation}
and we shall use the evolution equation of
\begin{equation}
(n-1)H\f^\frac12
\end{equation}
to express the Hamilton constraint. 
 
We immediately deduce
\begin{equation}
\begin{aligned}
(\f^\frac12)'&=\frac14 \f^\frac12 g^{ij}\dot g_{ij}\\
&=-\frac12 \f^\frac12 Hw
\end{aligned}
\end{equation}
\cf \fre{3.8}, and obtain, in view of \re{6.7.1} and \re{3.31.6}, 
\begin{equation}
\begin{aligned}
(n-1)(H\f^\frac12)'&=-(n-1)\tilde\D w \f^\frac12\\
&\q+(n-1)\{\abs A^2+\bar R_{\al\bet}\nu^\al\nu^\bet\}w \f^\frac12-\frac{n-1}2H^2\f^\frac12 w\\
&=-(n-1)\tilde\D w \f^\frac12+(n-1)(\abs A^2-H^2)\f^\frac12w\\
&\q+\frac{n-1}2H^2\f^\frac12 w+(n-2)\{\ein-\Lam\}\f^\frac12w\\
&\q -2\Lam \f^\frac12 w.
\end{aligned}
\end{equation}

Employing now the Hamilton condition and observing that
\begin{equation}
\frac12\{\abs A^2 -H^2-(R-2\Lam)\}=-\{\ein -\Lam\},
\end{equation}
\cf \cite[equ. (1.1.43)]{cg:cp}, we conclude that the evolution equation
\begin{equation}
\begin{aligned}
(n-1)(H\f^\frac12)'&=-(n-1)\tilde\D w \f^\frac12+(n-1)(R-2\Lam)\f^\frac12 w\\
&\q-2\Lam \f^\frac12 w+\frac{n-1}2H^2\f^\frac12 w
\end{aligned}
\end{equation}
is equivalent to the Hamilton condition provided the tangential Einstein equations are valid.

Finally, expressing the time derivative on the left-hand side by the Poisson brackets such that
\begin{equation}\lae{6.22.1}
\begin{aligned}
(n-1)\{H\f^\frac12,\mc H\}&=-(n-1)\tilde\D w \f^\frac12+(n-1)(R-2\Lam)\f^\frac12 w\\
&\q-2\Lam \f^\frac12 w+\frac{n-1}2H^2\f^\frac12 w
\end{aligned}
\end{equation}
we conclude that the Hamilton equations and the geometric evolution equation \re{6.22.1} are equivalent to the full Einstein equation, \cf the proof of \frt{3.1}.

Switching to the gauge $w=1$ we then quantize the equation \re{6.22.1}. Because of the relation \re{6.16.1} the left-hand side of \re{6.22.1} is transformed to
\begin{equation}
i[\hat H,\f^{-\frac12}\hat g_{ij}\hat\pi^{ij}]=[\hat H,\f^{-\frac12}g_{ij}\frac\de{\de g_{ij}}],
\end{equation}
where $\hat H$ is the transformed Hamiltonian. On the other hand,
\begin{equation}\lae{6.25.1}
\begin{aligned}
\f^{-\frac 12}g_{ij}\frac{\de}{\de g_{ij}}&=\sqrt{n(n-1)}\nu^a D_a=\sqrt{n(n-1)}\nu^0 D_0\\
&=\frac n4\frac{\pa}{\pa t},
\end{aligned}
\end{equation}
where $\nu^a$ is the future unit normal of the hypersurfaces 
\begin{equation}
M(t)=\{\xi^0=t\},
\end{equation}
i.e., the left-hand side of \re{6.25.1} is a constant multiple of the covariant derivative with respect to $t$ in the fiber when the differential operator is applied to functions $u=u(x,g_{ij})$. Hence,
\begin{equation}
\begin{aligned}
&\msp[100]\hp{}[\hat H,\f^{-\frac 12}g_{ij}\frac{\de}{\de g_{ij}}] u\\
&=\f^{-\frac12}g_{ij}\frac\de{\de g_{ij}}\{(R-2\Lam)u\f\}-\f^{-\frac12}(R-2\Lam)\f g_{ij}\pde u{g_{ij}}\\
&= \f^{-\frac12}\{\frac n2(R-2\Lam) u\f -Ru\f -(n-1)\tilde \D u \f\},
\end{aligned}
\end{equation}
in view of \fre{4.47}. The transformation of the right-hand side of \re{6.22.1}, note that $w=1$, yields
\begin{equation}
(n-1) (R-2\Lam)u\f^{\frac12} -2\Lam u\f^\frac12+\f^\frac12\frac{n-1}2H^2 u,
\end{equation}
where
\begin{equation}
\begin{aligned}
\f^\frac12\frac{n-1}2H^2 u&= -\frac n2\f^{-\frac12}\{\frac1{n(n-1)}\f^{-1} g_{ij}g_{kl}\frac \de{\de g{ij}}\frac \de{\de g_{kl}}\}u\\
&= -\frac n2\f^{-\frac12}(\nu^a\nu^bD_aD_b u)
\end{aligned}
\end{equation}
or
\begin{equation}
\f^\frac12\frac{n-1}2H^2 =-\frac n2\f^{-\frac12}D_a(\nu^a\nu^bD_b u)
\end{equation}
depending on the ordering of the  derivatives.

Observing that
\begin{equation}
\nu=(\nu^0,0,\ldots,0)
\end{equation}
and
\begin{equation}
\nu^0=\frac14\sqrt{\frac n{n-1}}
\end{equation}
we obtain, after multiplying both sides with $\f^\frac12$, the hyperbolic equations
\begin{equation}
\frac1{32}\frac {n^2}{n-1}\Ddot u -(n-1)t^2\tilde\D u  -\frac n2 Rt^2u+n\Lam t^2u=0
\end{equation}
or
\begin{equation}
\frac1{32}\frac {n^2}{n-1}t^{-m}\frac{\pa}{\pa t}(t^m\dot u) -(n-1)t^2\tilde\D u  -\frac n2 Rt^2u+n\Lam t^2u=0
\end{equation}
where we recall that $\f=t^2$, \cf \re{4.58} and \fre{4.74}.
 
 These equations can be rewritten, as before, by observing that
 \begin{equation}
g_{ij}=t^{\frac4n}\s_{ij},
\end{equation}
such that
\begin{equation}
\tilde \D u=t^{-\frac4n}\tilde \D_{\s_{ij}}u
\end{equation}
and
\begin{equation}
R=t^{-\frac4n}R_{\s_{ij}},
\end{equation}
where $R_{\s_{ij}}$ is the scalar curvature of the metric $\s_{ij}$. Both equations are hyperbolic equations in $E$, where $u=u(x,t,\xi^A)$, $1\le A\le m$, and $\s_{ij}=\s_{ij}(x,\xi^A)$. However, for fixed $(\xi^A)$, we may consider these equations as hyperbolic equations in
\begin{equation}
\so\times \R[*]_+,
\end{equation}
where the solutions as well as the metric depend on an additional parameter $(\xi^A)$. To simplify the notation let us drop the tilde over the Laplacian and stipulate that the Laplacian as well as the scalar curvature refer to the metric $\s_{ij}$. Then we can rewrite the equations as
\begin{equation}\lae{6.15}
\frac1{32}\frac{n^2}{n-1}\Ddot u
-(n-1) t^{2-\frac4n}\D u-\frac {n}2 t^{2-\frac4n}Ru+nt^2\Lam u=0.
\end{equation}
and
\begin{equation}\lae{6.14}
\begin{aligned}
\frac1{32}\frac{n^2}{n-1}t^{-m}\pde{(t^m\dot u)}t
-(n-1) t^{2 -\frac4n}\D u-\frac {n}2t^{2-\frac4n}Ru+nt^2\Lam u=0
\end{aligned}
\end{equation}
We also note that 
\begin{equation}\lae{6.41.1}
\det \s_{ij}=\det\chi_{ij}
\end{equation}
and that $\s_{ij}\in E_1$ is arbitrary but fixed.
\bl
Both operators are symmetric with respect to the Lorentzian metric
\begin{equation}\lae{6.17}
d\bar s^2=-\frac{32(n-1)}{n^2}dt^2+\s_{ij}dx^idx^j
\end{equation}
and they are normally hyperbolic with respect to the metric
\begin{equation}\lae{6.18}
d\tilde s^2=-\frac{32(n-1)}{n^2}dt^2+\frac1{n-1}t^{\frac4n-2}\s_{ij}dx^idx^j.
\end{equation}
\el
Thus, if 
\begin{equation}
Q=\so\times \R[*]_+
\end{equation}
is globally hyperbolic with respect to these metrics, and if we denote $Q$ equipped with the metric \re{6.18} by $\tilde Q$ and stipulate that $Q$ is equipped with the metric \re{6.17}, then the results from \rs{4} and \rs{5} can be applied to the present setting. 
\bl
Assume that the metric
\begin{equation}
\s_{ij}(x,\xi)\in E_1,
\end{equation}
where $\xi=(\xi^A)$ is fixed, is complete, then the Lorentzian manifolds $Q$ and $\tilde Q$ are globally hyperbolic, and the hypersurfaces
\begin{equation}
M_\tau=\{t=\tau\}\su Q
\end{equation}
are Cauchy hypersurfaces.
\el
\bp
Let us only consider $Q$. From the proof of \frl{4.2} we infer that the claims are correct if a bounded curve
\begin{equation}
\ga(s)\su \so,\qq s\in I,
\end{equation}
where bounded means, bounded relative to $\s_{ij}$, is relatively compact which is the case, if $(\so,\s_{ij})$ is complete.
\ep

In the next theorem we would like to prove that the solutions depend smoothly on $\xi$. In order to achieve this, the Cauchy values have to be prescribed on $E_1(\tau)$ and not only on $M_\tau$.

\bt
Let $P$ be one of the hyperbolic operators in \re{6.14} or \re{6.15}, and let $E_1(\tau)$ be given as well as functions $f\in C^\un_c(E)$ and $u_0,u_1\in C^\un_c(E_1(\tau))$. These functions depend on $(x,t,\xi)$. Since $f,u_0,u_1$ have compact support, the corresponding $\xi$, such that $f(\xi), u_0(\xi), u_1(\xi)$ do not identically vanish in $Q$, are contained in a relatively compact, open set $U$. Assume  that the metrics
\begin{equation}
\s_{ij}(x,\xi),\q\xi\in U,
\end{equation}
are all complete, then, the Cauchy problems
\begin{equation}
\begin{aligned}
Pu&=f\\
\fv u{E_1(\tau)}&=u_0\\
\fv{\dot u}{E_1(\tau)}&=u_1
\end{aligned}
\end{equation}
are uniquely solvable in $(Q,\s_{ij})$ for all $\xi\in U$ such that
\begin{equation}
u=u(x,t,\xi)\in C^\un (\fv EU),
\end{equation}
where
\begin{equation}
\fv EU=\{(x,t,\xi): \xi\in U\}.
\end{equation}
\et
\bp
First, we apply the results in \rs{4} to the operator $P$ and the globally hyperbolic spaces $Q$ and $\tilde Q$ for each $\xi\in U$ to conclude that, for fixed $\xi\in U$, the solutions exist, are uniquely determined, and are smooth in $(x,t)$. Arguing then as in the proof of \cite[Theorem 5.4]{cg:qgravity}, where we considered solutions of hyperbolic problems in the fibers of $E$, where the solutions and the data were depending on the parameter $x\in\so$, we can prove, by considering the problems in $\tilde Q$, so that $P$ is normally hyperbolic, that the solutions are also smooth in $\xi$. Moreover, for each $\xi\in U$, the solution $u(\xi)$ satisfies the known support properties of solutions in $\tilde Q$.
\ep
The equations \re{6.14} or \re{6.15} can be looked at as being gravitational wave equations and the solutions $u=u(x,\xi)$ can be considered to be gravitons. Note that $\xi=(\xi^A)$ are coordinates for the metrics in the fibers, and the pair $(x,\xi)$ represents the metric $\s_{ij}(x,\xi)$ in $\so$.

If $\so$ is compact then we shall construct variational solutions of equation \re{6.15} with finite energy which may be considered to provide a spectral resolution of the problem for fixed $\xi$.

Let us start with the following well-known lemma:
\bl
Let $\so$ be compact equipped with the metric $\s_{ij}=\s_{ij}(\xi)$. Then the eigenvalue problem
\begin{equation}\lae{6.27}
-(n-1)\D v-\frac{n}2R v=\mu v
\end{equation}
has countably many solutions $(v_i,\mu_i)$ such that
\begin{equation}
\mu_0<\mu_1\le \mu_2\le\cdots,
\end{equation}
\begin{equation}
\lim_i\mu_i=\un,
\end{equation}
and
\begin{equation}
\int_\so\bar v_iv_j=\de_{ij},
\end{equation}
where we now consider complex valued functions. The eigenfunctions are a basis for $L^2(\so,\Cc)$ and are smooth.
\el

Now we argue similarly as in \cite[Subsection 6.7]{cg:qfriedman}: Choose any eigenfunction $v=v_i$ with positive eigenvalue $\mu=\mu_i$, then we look at solutions $u$ of \re{6.15} of the form
\begin{equation}
u(x,t)=w(t)v(x).
\end{equation}
Inserting $u$ in the equation we deduce
\begin{equation}
\frac1{32}\frac{n^2}{n-1}\Ddot w +\mu t^{2-\frac4n}w+nt^2\Lam w=0,
\end{equation}
or equivalently,
\begin{equation}\lae{6.33}
-\frac1{32}\frac{n^2}{n-1}\Ddot w -\mu t^{2-\frac4n}w-nt^2\Lam w=0.
\end{equation}
This equation can be considered to be an implicit eigenvalue problem with eigenvalue $\Lam$.

To solve \re{6.33} we first solve 
\begin{equation}\lae{6.34}
-\frac1{32}\frac{n^2}{n-1}\Ddot w +nt^2w=\lam\mu t^{2-\frac4n}w,
\end{equation}
where $\lam$ is the eigenvalue. Let $I=\R[*]_+$ and $H$ be the embedded  subspace of the Sobolev space $H^{1,2}_0(I)$
\begin{equation}
H\hra H^{1,2}_0(I,\Cc)
\end{equation}
defined as the completion of $C^\un_c(I,\Cc)$ under the norm of the scalar product
\begin{equation}
\spd w{\tilde w}_1=\int_I\{\bar w'\tilde w' +t^2\bar w\tilde w\},
\end{equation}
where a prime or a dot denotes differentiation with respect to $t$. Moreover, let $B$, $K$ be the symmetric forms
\begin{equation}
B(w,\tilde w)=\int_I\{\frac1{32} \frac{n^2}{n-1}\bar w'\tilde w'+nt^2\bar w\tilde w\}
\end{equation}
and
\begin{equation}
K(w,\tilde w)=\int_I\mu t^{2-\frac4n}\bar w\tilde w,
\end{equation}
then the eigenvalue equation \re{6.34} is equivalent to 
\begin{equation}\lae{6.39}
B(w,\f)=\lam K(w,\f)\qq\A\,\f\in H
\end{equation}
as one easily checks.
\bl
The quadratic form $K(w)=K(w,w)$ is compact relative to the quadratic form $B$, i.e., if $w_k\in H$ converges weakly to $w\in H$
\begin{equation}
w_k\rha w\qq \text{in } H,
\end{equation}
then
\begin{equation}
K(w_k)\ra K(w).
\end{equation}
\el
\bp
The proof is essentially the same as the proof of \cite[Lemma 6.8]{cg:qfriedman} and will be omitted.
\ep

Hence the eigenvalue problem \re{6.39} has countably many solutions $(\tilde w_i,\lam_i)$ such that
\begin{equation}\lae{6.42}
0<\lam_0<\lam_1<\cdots,
\end{equation}
\begin{equation}
\lim\lam_i=\un
\end{equation}
and
\begin{equation}
K(\tilde w_i,\tilde w_j)=\de_{ij}.
\end{equation}
For a proof of this well-known result, except the strict inequalities in \re{6.42},  see e.g. \cite[Theorem 1.6.3, p. 37]{cg:pdeII}. Each eigenvalue has multiplicity one since we have a linear ODE of order two and all solutions satisfy the boundary condition 
\begin{equation}\lae{6.45}
\tilde w_i(0)=0.
\end{equation}
The kernel is two-dimensional and the condition \re{6.45} defines a one-dimen\-sional subspace. Note, that we considered only real valued solutions to apply this argument. 

Finally, the functions
\begin{equation}
w_i(t)=\tilde w_i(\lam_i^{-\frac n{4(n-1)}}t)
\end{equation}
then satisfy \re{6.33} with eigenvalue
\begin{equation}
\Lam_i=-\lam_i^{-\frac n{n-1}}
\end{equation}
and 
\begin{equation}
u_i=w_iv
\end{equation}
is a solution of the wave equation \re{6.15} with finite energy
\begin{equation}\lae{6.49} 
\begin{aligned}
\norm{u_i}^2=\int_Q\{\abs {\dot u}^2+(1+t^2)\s^{ij}\bar u_iu_j+\mu t^{2-\frac4n}\abs u^2\}<\un.
\end{aligned}
\end{equation}
Note that the actual energy is defined by a weaker norm
\begin{equation}
\int_Q\{\abs {\dot u}^2+t^{2-\frac4n}\s^{ij}\bar u_iu_j+\mu t^{2-\frac4n}\abs u^2\}
\end{equation}
which is of course bounded too.

Let us summarize these results:
\bt\lat{6.7}
Assume $n\ge 2$ and $\so$ to be compact and let $(v,\mu)$ be a solution of the eigenvalue problem \re{6.27} with $\mu>0$, then there exist countably many solutions $(w_i,\Lam_i)$ of the implicit eigenvalue problem \re{6.33} such that
\begin{equation}
\Lam_i<\Lam_{i+1}<\cdots <0,
\end{equation}
\begin{equation}
\lim_i\Lam_i=0,
\end{equation}
and such that the functions
\begin{equation}
u_i=w_i v
\end{equation}
are solutions of the wave equations \re{6.15}. The transformed eigenfunctions
\begin{equation}
\tilde w_i(t)=w_i(\lam_i^{\frac n{4(n-1)}}t), 
\end{equation}
where
\begin{equation}
\lam_i=(-\Lam_i)^{-\frac{n-1}n}
\end{equation}
form a basis of the Hilbert space $H$ and also of $L^2(\R[*]_+,\Cc)$.
\et

\br
Let $\s_{ij}$ be a smooth and complete Riemannian metric in $\so$, then $\s_{ij}$ is in general only a section of $E$ but not an element. However, the metric $\chi_{ij}$ in \fre{4.1}, which we used to define $\f$ in order to  replace the density $\sqrt g$, can certainly be assumed to belong to $E$, and hence to the subbundle $E_1$, because we can easily define a covering of local trivializations where $\chi$ is always part of the generating local frames. Since $\chi$ is chosen arbitrarily we may just as well assume that
\begin{equation}
\chi_{ij}=\s_{ij}.
\end{equation}
Hence, the hyperbolic equations \re{6.15} or \re{6.14}, which are supposed to describe a model for quantum gravity, can be applied to any given smooth and complete metric metric $\s_{ij}$, or more precisely, to any complete Riemannian manifold $(\so,\s_{ij})$.
\er
Let us formulate this result  in case of equation \re{6.15} as a theorem:
\bt
Let $(\so,\s_{ij})$ be a connected, smooth and complete  $n$\ndash di\-mensional Riemannian mani\-fold   and let 
\begin{equation}
Q=\so\times \R[*]_+
\end{equation}
be the corresponding globally hyperbolic spacetime equipped with the Lorentzian metric \re{6.17} or, if necessary, with \re{6.18}, then the hyperbolic equation
\begin{equation}
\frac1{32}\frac{n^2}{n-1}\Ddot u
-(n-1) t^{2-\frac4n}\D u-\frac {n}2 t^{2-\frac4n}Ru+nt^2\Lam u=0,
\end{equation}
where the Laplacian and the scalar curvature correspond to the metric $\s_{ij}$, describes a model of quantum gravity. If $\so$ is compact a spectral resolution of this equation has been proved in \rt{6.7}.
\et
\br
If $\so$ is not compact, then we proved in \cite{cg:uf4,cg:qbh,cg:qbh2} that a spectral resolution is possible if either $\so$ is an asymptotically Euclidean Cauchy hypersurface of a globally hyperbolic spacetime $N$, or, if $N$ is a black hole, if $\so$ is the smooth limit of Cauchy hypersurfaces representing the event horizon though with a different metric.
\er
\br
When $\s_{ij}$ is the metric of a space of constant curvature then the equation \re{6.15}, considered only for functions $u$ which do not depend on $x$,  is identical to the equation obtained by quantizing the Hamilton constraint in a Friedmann universe without matter but including a cosmological constant. The equation is the ODE
\begin{equation}
\frac1{16} \frac n{n-1} \Ddot u-R r^{2-\frac 4n}u+2r^2\Lam u=0,\qq 0<r<\un,
\end{equation}
\cf \cite[equ. (3.37)]{cg:qfriedman}, though the equation there looks differently, since in that paper we divided the Lagrangian by $n(n-1)$.
\er



%\backmatter
%\includepdf[pages=-]{/Users/claus/Documents/Scanned-Documents/}
\bibliographystyle{hamsplain}
%\bibliography{mrabbrev,publications}
\providecommand{\bysame}{\leavevmode\hbox to3em{\hrulefill}\thinspace}
\providecommand{\href}[2]{#2}


%\listoffigures


%\cleardoublepage

%\thispagestyle{empty}
%\closegraphsfile
\end{document}